

Università degli Studi di Padova



Università degli Studi di Padova

Dipartimento di Ingegneria dell'Informazione

ICT for internet and multimedia

A META-ANALYSIS OF SECURITY PROBLEMS IN COMMUNICATION NETWORKS APPROACHED THROUGH GAME THEORY

Supervised By:

Prof. Leonardo Badia

Presented By: *Iyad Zabalawi*

ACKNOWLEDGMENT

I would like to express my deepest gratitude to the University of Padova, specifically to the Department of Information Engineering, for providing an enriching environment for my academic and personal growth. The facilities, resources, and most importantly, the community, have played a pivotal role in my journey through the master's program.

Special thanks to Professor Leonardo Badia, whose guidance, support, and expertise were invaluable throughout the research and writing of this thesis. Professor Badia's insights and encouragement have not only helped me to navigate the challenges of my study but have also inspired me to pursue excellence in my work with dedication and integrity.

I am profoundly grateful to my family, whose unwavering love, support, and belief in my abilities have been my stronghold. Their sacrifices, understanding, and encouragement have been the bedrock of my resilience and success. To my parents, whose dreams for me have always been bigger than those I dreamt for myself, thank you for everything.

My heartfelt appreciation also goes out to my friends, both within and outside the University, for their companionship, laughter, and invaluable support. Their presence has made this journey not only bearable but also enjoyable, providing much-needed breaks from the rigors of academic life and enriching my university experience in countless ways.

Lastly, I wish to extend my gratitude to everyone who directly or indirectly contributed to my thesis. Your input, feedback, and moral support have been essential to my growth and the completion of this project.

ABSTRACT

Security is a paramount challenge in communication networks to ensure the integrity and confidentiality of transmitted data. This thesis presents a meta-analysis discussing the literature investigating security issues that are pervasive in communication networks, employing a game theory framework for a comprehensive examination. By synthesizing existing research, the study assesses various security problems, including but not limited to eavesdropping, denial of service attacks, and node compromise. The application of game theory provides a strategic lens to understand the dynamics between network participants, attackers, and defenders. This can contribute to the identification of possibly underestimated security threats as well as a deeper understanding of confidentiality and privacy paradigms in communication networks, fostering the development of robust and adaptive security solutions in the face of evolving cyber threats.

TABLE OF CONTENTS

LIST OF FIGURESII					
LIST OF TABLES III					
LIST OF ACRONYMS/ABBREVIATIONSIV					
1	INTRODUCTION				
1.1	motivation	8			
1.2	1.2 contribution				
1.3	Outline	9			
2	BACKGROUND	. 10			
2.1	2.1 The Problem of Security in the Digital Age 10				
2.2	Game Theory and its Relevance to Networks	.11			
2.3	Security and Game Theory Combination	. 12			
3	REVIEW				
2 1	Introduction	14			
J.I 2 7	Introduction	. 14			
3.2	3.2.1 Jamming Attacks and Network Security	15			
	3.2.1 Jamming Attacks and Information Security	31			
	3 2 3 Favesdropping and Information Security	42			
3.3	Security Problems Addressing Age of Information (AoI)	.52			
0.0	3.3.1 Jamming Attacks and Network Security.	.53			
	3.3.2 False Data Injection and Information Security	. 61			
	3.3.3 Eavesdropping and Information Security	. 68			
3.4	Conclusion	. 76			
4	RESULTS	.77			
4.1	Introduction	. 77			
4.2	Methodological Overview	.77			
	4.2.1 Categorization of Research Papers	. 77			
	4.2.2 Statistical Analysis				
4.3	Ouantitative Analysis	. 79			
	4.3.1 Distribution by Security Concern:	. 79			
	4.3.2 Game-Theoretic Approaches:	. 81			
	4.3.3 Application Areas:	. 84			
4.4	Trends and Patterns	. 85			
	4.4.1 Temporal Trends:	. 85			
4.5	Comparative Analysis	. 86			
	4.5.1 Effectiveness of Approaches:	. 86			
	4.5.2 Innovation and Impact	. 88			
4.6	Discussion	. 89			
	4.6.1 Synthesis of Statistical Analysis	. 89			
	4.6.2 Gaps in Current Research	. 89			
17	4.0.5 FOURINAL AREAS IOF FULLIFE WORK	.90			
4. /		, 71			
5	CONCLUSION	. 93			
6	REFERENCES				

LIST OF FIGURES

Figure 3-1: Sensor mixed strategy	
Figure 3-2: Jammer mixed strategy	
Figure 3-3 Flow of energy and data between different parts of smart grids	33
Figure 3-4 RSU and SV interaction in a VANET	
Figure 3-5 Payoff Surfaces	
Figure 3-6	64
Figure 3-7	64
Figure 3-8	65
Figure 3-9	65
Figure 3-10 P and Q at the NE for $C = 4$	65
Figure 3-11 P and Q at the NE for $K = 0.1$	65
Figure 4-1 Distribution by Security Concern.	79
Figure 4-2 zero-sum and non-zero-sum games.	81
Figure 4-3 complete and incomplete information games.	82
Figure 4-4 Simultaneous and Sequential games	82
Figure 4-5 Static and Dynamic games	83
Figure 4-6 Application Areas.	
Figure 4-7 Temporal Trends	85

LIST OF TABLES

Table 4-1: Distribution by Security Concern (General Security Issues).	. 80
Table 4-2: Distribution by Security Concern (AoI)	. 80

LIST OF ACRONYMS/ABBREVIATIONS

DoS	Denial of Service
IDS	Intrusion Detection Systems
IPS	Intrusion Prevention Systems
VANET	vehicular ad hoc network
SV	Smart Vehicle
RSU	Stationary Roadside Unit
UASN	Underwater Acoustic Sensor Network
WSN	Wireless Sensor Networks
SINR	Signal-to-Interference-plus-Noise Ratio
AoI	Age of Information
AoII	Age of Incorrect Information
SV	Smart Vehicle
RSU	Roadside Unit
FH	Frequency Hopping
BDD	Bad Data Detection
LP	Linear Programming
NE	Nash Equilibrium
PoA	Price of Anarchy
UAV	unmanned aerial vehicle
GCS	Ground Control Station
BS	Base Station
MCS	Mobile Crowd Sensing

Chapter One

1 INTRODUCTION

1.1 MOTIVATION

Network technologies have significantly improved our access to information and communication. However, they also present us with security challenges. We face various threats, including internet attacks, cybercrimes, Denial of Service (DoS) attacks, unauthorized data access, and information theft. These issues can lead to financial losses, data breaches, and reputational damage for both public institutions and private companies. The continuous emergence of new hackers, cybercrimes, and security incidents [1], [2], [3] highlights the complexity of network security.

The conventional approaches to safeguarding networks are no longer adequate. Traditionally, security measures have relied on preventive tools like firewalls or reactive solutions such as antivirus software. The increasing sophistication of attacks has made Intrusion Detection Systems (IDSs) a critical addition to organizational security infrastructures [4]. IDSs monitor and analyze network activities to identify potential threats using techniques like signature identification, pattern detection, and statistical analysis [5]. Upon detecting an attack, they notify administrators who can then take action to mitigate the threat. Some IDSs, known as Intrusion Prevention Systems (IPSs), can even respond to attacks without administrator intervention [6]. Despite their utility, IDSs face limitations in dealing with advanced, organized threats due to their reliance on simple, ad hoc methods [7].

Game theory has been proposed as a means to enhance network security. Traditional security solutions often lack a quantitative decision-making framework [8]. Game theory examines scenarios where individuals or entities with conflicting objectives interact, providing a mathematical basis for analyzing and modeling network security issues. It enables the consideration of numerous potential outcomes before making strategic decisions [9]. This approach can significantly refine the decision-making process for network administrators.

Moreover, the assessment of network security [10], evaluating aspects such as confidentiality, integrity, availability, vulnerabilities, and overall risk, is crucial. Security measurement encompasses a broad range of considerations, including risk assessment [11]. These measurements take into account the dynamic interplay between attackers and defenders, which can influence the outcome. The likelihood of an attack on a network, for example, requires an understanding of both the attackers' and defenders' strategies. As this interaction resembles a game, game theory is ideally suited to forecasting attacker behavior and guiding defender responses. Consequently, game theory-based strategies have been suggested to address network security challenges.

1.2 CONTRIBUTION

This thesis reviews the game theoretic approaches that have been used to address security problems in networks. It categorizes these approaches based on the types of attacks and the models of games that they consider. This thesis aims to evaluate the strengths and weaknesses of different game theory approaches, to identify the gaps and challenges in the existing literature, and to suggest new avenues for research on network security problems.

1.3 OUTLINE

The rest of this thesis is as follows: Chapter II introduces the general problem of security and the basic concepts and relevance of game theory for networks; Chapter III presents a review of the papers that apply game theory to network security, organized by different criteria such as game types, game settings, and game objectives; Chapter IV analyses the results of the papers systematically, showing the frequency and distribution of various methods and outcomes; Chapter V concludes the thesis.

Chapter Two

2 BACKGROUND

2.1 THE PROBLEM OF SECURITY IN THE DIGITAL AGE

As computer and internet technologies continue to evolve, cybercrime has emerged as a major challenge for law enforcement agencies worldwide [29]. Unlike traditional crimes that occur in the physical world, cybercrimes take place online or in digital environments. This distinction does not mean that cyber and physical crimes are entirely separate; in fact, they often intersect [30].

In our physical society, despite the presence of laws, regulations, and law enforcement agencies aimed at reducing crime, it is understood that crimes cannot be entirely eliminated. Similarly, in the digital world, cybercrimes are seen as inevitable. However, the legal framework for dealing with cybercrimes is still developing. The rapid advancement of technology and a lack of comprehensive understanding of cybercrimes have made it difficult to establish strong legal protections for internet users [31]. This gap in legal infrastructure makes it challenging to apprehend and prosecute cybercriminals effectively and to gather the necessary evidence for conviction.

One of the unique aspects of cybercrime is that its effects are often not immediately noticeable. Unlike physical crimes, which might result in visible harm or loss, the damage from cyber attacks—such as data breaches or theft—can go undetected for long periods. Cybercriminals exploit the interconnected nature of digital networks, allowing them to launch attacks from anywhere in the world with relative ease, a feat that would be significantly more challenging in the physical world [32].

Initially, the focus of digital technology development was on enhancing functionality, like information processing and communication, rather than securing these processes against threats. As a result, the measures for preventing, detecting, and responding to cyber threats are not as advanced as they should be [33]. This makes digital environments particularly vulnerable to criminal activities.

Cybercrime encompasses a wide range of illegal activities that exploit information technology, including hacking, data theft, electronic fraud, and identity theft [12]. The impact of cybercrime is substantial and growing, posing serious risks to individuals and organizations alike. A study highlighted the alarming frequency of cybercrime in Britain, with a new incident occurring every 10 seconds and over three million attacks recorded in 2006 alone [13]. The digital nature of these crimes allows many to go unreported, with estimates suggesting that up to 90% of cybercrimes are not brought to the attention of authorities [34].

Addressing cybercrime requires ongoing efforts from researchers and law enforcement to develop more effective strategies for protecting digital spaces and ensuring the security of all users [35].

2.2 GAME THEORY AND ITS RELEVANCE TO NETWORKS

Game theory is a way of using math to study and solve problems that involve players making decisions. It looks at different parts of a game, like who is playing (players), the steps they take (actions), what they know when they make a move, and what they get out of it (payoffs). It is important to remember that game theory believes all players look to make the best choice for themselves [36].

A game is any situation where two or more players interact. Games can be competitive, where players work against each other, or cooperative, where they work together. In game theory, we focus on four main ideas:

- Players: These are the ones taking part in the game. They could be people, groups, or even animals, as long as they can interact [37].
- Actions: What a player decides to do at any point in the game. Players are assumed to know all possible actions they and others can take.
- Payoffs: This is what players end up with after the game, which could be good or bad. It is the result of the actions they have taken [38].
- Strategies: A plan a player follows during the game, deciding what to do based on what is happened so far. Strategies can be straightforward or complex [39].

In game theory, since players are seen as rational, they pick strategies that they think will give them the best payoff, especially in response to what others are doing. This leads to the idea of "Equilibrium," where players find the best strategy they can, considering others' strategies. One specific type of equilibrium is the "Nash Equilibrium [14]" which is a set of strategies where no player can benefit by changing their strategy alone. This concept is useful for finding solutions in games.

Game theory can answer the question regarding how the defender will react to the attacker, and vice versa, in cyber security. The strategic interaction between them is captured by a two-player game in which each player attempts to maximize his or her own interests. The attacker's strategy depends heavily on the defender's actions and vice versa. Thus, the effectiveness of a defense mechanism relies on both of the defender's and attacker's strategic behaviors. Using the game-theoretic approach, tactical analysis is performed to investigate the attack from a single node or multiple nodes [40]. Hence, game theory is useful to investigate the strategic decision-making situations of the defender and/or to analyze the incentives of the attackers.

2.3 SECURITY AND GAME THEORY COMBINATION

When we mix the ideas of security with game theory, we create a strong way to predict and fight against online threats. This mix uses game theory's ability to forecast how cyber attackers and defenders might act [41]. Seeing cyber security as a kind of game helps defenders guess the attackers' moves better, plan their defenses more smartly, and come up with stronger ways to stop them [42]. This method means security can keep up with and respond to the changing tricks of cyber attackers.

Using game theory in cyber security means looking at how different security actions might pay off, considering what attacks could happen. This helps find weak spots in computer systems and plan out defenses that either stop attacks before they happen or lessen their damage [43]. For example, security experts can use game theory to imagine different attack situations and see how well their defenses would work. This helps them make smart choices about where to put their efforts and resources to protect against cyber threats [44].

Also, game theory shows how connected the world of cybersecurity is. It demonstrates that what one person or company does affects everyone else's safety, emphasizing the need for teamwork in fighting cyber threats [45]. This way, security is not just one person's job, but a team effort that makes the whole network stronger.

In short, bringing together security and game theory not only makes it easier to understand and fight cyber threats but also leads to creating new and adaptable ways to defend against them [46]. By looking at cyber security through the lens of game theory, security experts and companies can be better prepared for the clever tactics of cyber attackers, making the digital world a safer place for everyone.

Chapter Three

3 REVIEW

3.1 INTRODUCTION

Within the larger framework of our study on the security of communication networks, this chapter focuses on the innovative role of game theory in tackling security threats. As we explore the challenges of protecting these networks, it becomes evident that traditional security methods often fall short against the complex and evolving nature of cyber threats. These issues, including jamming attacks, false data injection, and eavesdropping, threaten the integrity, availability, and confidentiality of data and challenge the core trust in communication networks.

Therefore, this chapter aims to contribute to our research by examining how game theory can be applied to develop more effective security strategies. Game theory, which analyzes the strategic interactions between rational players, is a valuable tool for understanding the competitive dynamics in network security. It provides a framework for identifying the best defense tactics in the face of threats, encouraging a proactive and strategic response to network protection.

The chapter is structured to build upon these ideas logically. It starts with Section 3.2, which introduces the general security issues that game theory helps address, setting the stage for a deeper analysis. The following sections delve into specific threats: jamming attacks (Section 3.2.1), false data injection (Section 3.2.2), and eavesdropping (Section 3.2.3), discussing each through the lens of game theory and outlining potential strategic counteractions.

Expanding the discussion, Section 3.3 looks at how these security challenges affect the Age of Information (AoI), an essential measure of how current and relevant information is within a network. This section reexamines the earlier threats in the context of their impact on AoI and the strategic measures necessary to mitigate these effects.

In conclusion, Section 3.4 brings together the insights gained, highlighting the crucial role of game theory in creating strong and adaptive security strategies. As part of our

wider research, this chapter emphasizes the need for a strategic, informed, and evolving approach to defending against the diverse range of threats facing modern communication networks.

3.2 GENERAL SECURITY ISSUES ADDRESSED THROUGH GAME THEORY

3.2.1 Jamming Attacks and Network Security

In the realm of network security, jamming represents a formidable challenge to safeguarding information. This type of attack, aimed deliberately at hindering network functionality, falls under the category of denial of service (DoS) attacks. By interfering with transmissions, jamming attacks block legitimate users from accessing communication channels.

The dynamics of jamming can be understood through a game-theoretical perspective, involving two main participants: the jammer and the communicator (a duo of transmitter and receiver), each with opposing goals. The jammer's role is to obstruct and deny access to the wireless channel for regular users by disrupting their communications. On the other hand, the communicator nodes strive to efficiently use the wireless channel to maximize their data throughput. This scenario can also extend to a game between the jammer and a monitor node, where the monitor's role is to identify and counteract the jamming attacks.

Game Theory-Based Anti-Jamming Strategies for Frequency Hopping Wireless Communications [27]

Frequency hopping (FH) wireless communications have traditionally been employed to counteract these threats by rapidly switching frequencies in a pattern known to both the sender and receiver. However, the dynamic and adaptive nature of modern jamming techniques necessitates more sophisticated countermeasures. This presents a novel approach to this problem by applying game theory to develop anti-jamming strategies that enhance the resilience of frequency hopping wireless networks.

Game Theory Approach

The paper introduces a bimatrix game framework to model the interactions between a transmitter (user) and a jammer as a strategic game. Game theory, a mathematical framework designed to analyze strategic interactions among rational decision-makers, is utilized to understand and predict the outcomes of this adversarial engagement. The authors detail the process of establishing a game model wherein both the transmitter and jammer have a set of strategies, and their choices determine the payoff for both parties.

In this context, the primary objective is to find the Nash equilibrium (NE) - a set of strategies where no player can benefit by unilaterally changing their strategy if the other's strategies remain unchanged. The paper meticulously derives the sufficient and necessary conditions for reaching the NE in the game, providing a foundational strategy for the transmitter to minimize the impact of jamming.

Here is a breakdown of the game in terms of its type, players, strategies, and payoffs:

Type of Game:

The game is formulated as a bimatrix game, which falls under the category of noncooperative games. In this context, it is a strategic game where each player knows the payoff functions of all players and makes their decision independently, aiming to maximize their own payoff without cooperation with the other player.

Players:

- Transmitter (User): The player trying to communicate information over the network without interference. The transmitter's goal is to maintain effective communication despite potential jamming.
- Jammer: The adversarial player attempting to disrupt the communication by jamming the frequencies used by the transmitter. The jammer's goal is to degrade the communication quality or prevent it altogether.

Strategies:

• Transmitter Strategies: The set of actions available to the transmitter includes choosing from a range of frequencies for hopping to evade jamming. The

strategy can also involve selecting transmission power levels and hopping patterns that are least likely to be predicted or affected by the jammer.

 Jammer Strategies: The jammer chooses from a set of frequencies to jam, aiming to target the frequencies that the transmitter is most likely to use. Additionally, the jammer decides on the power levels and patterns of jamming to effectively disrupt the transmitter's communication.

Payoffs:

- Transmitter Payoff: The transmitter's payoff is a function of successful communication rate, which depends on the ability to evade jamming and maintain signal integrity. High payoff is achieved when the transmitter can effectively communicate without interference by the jammer.
- Jammer Payoff: The jammer's payoff is inversely related to the transmitter's communication success. A higher payoff is achieved when the jammer successfully disrupts the communication, leading to poor signal quality or complete loss of communication for the transmitter.

Nash Equilibrium and Optimal Strategies

By exploring the linear constraints and mathematical formulations that govern the game, the authors are able to identify the conditions under which the Nash equilibrium can be achieved. This equilibrium represents a state where both the jammer and the transmitter's strategies are optimized against each other, making it impossible for either party to improve their outcome by changing their strategy alone.

Moreover, the paper bridges the gap between the Nash equilibrium strategy and the global optimal solution of the corresponding quadratic programming problem. This connection is crucial for understanding the theoretical optimal performance achievable in the presence of jamming and forms the basis for the proposed anti-jamming strategies.

Numerical Investigations and Performance Improvement

The theoretical framework is complemented with numerical investigations to validate the effectiveness of the proposed game-theoretic anti-jamming strategies. Through these simulations, the paper demonstrates a significant improvement in the performance of frequency hopping wireless communications under jamming attacks. This empirical evidence underlines the practical applicability and potential of game theory in designing robust anti-jamming mechanisms for real-world communication systems.

Conclusion

The paper is a foundational work in wireless communication security, highlighting game theory's role in countering jamming attacks. It establishes a framework for identifying optimal anti-jamming strategies via Nash equilibrium, laying groundwork for future secure communication research. This study, along with similar investigations [64][65], advances theoretical and practical aspects of anti-jamming mechanisms, promoting the resilience of wireless networks against complex threats. These collaborative efforts underline the significance of game-theoretic approaches in enhancing communication security.

A Game of One/Two Strategic Friendly Jammers Versus a Malicious Strategic Node [16]

This study offers a significant contribution to the understanding of security in communication networks through a game-theoretic lens. It specifically addresses the dynamics between friendly jammers and a malicious node in a scenario where unauthorized radio transmissions are a threat. The research stands out by employing a game-theoretical framework to dissect the interactions between these entities, providing insights that are both counterintuitive and pivotal for enhancing network security.

Game-Theoretical Framework

At the heart of the paper analysis is the strategic game between the jammers and the malicious node. The study assumes that the malicious node is rational, aiming to conduct unauthorized transmissions without being disrupted. Conversely, the friendly jammers' objective is to thwart these communications effectively. The game is modeled considering various factors, including the number of jammers, their coordination, and the strategic responses from the malicious node.

The analysis delves into two primary scenarios: a single jammer and multiple jammers acting without coordination. This distinction is crucial as it uncovers the nuances of jamming effectiveness and the strategic considerations each party must account for. The authors utilize a mathematical approach to explore the equilibrium states of these scenarios, shedding light on the optimal strategies for both the jammers and the malicious node. Here is an outline of the static (one-shot) game of complete information:

Type of Game

The game is a non-cooperative game, where players make independent decisions to maximize their individual payoffs without collaboration. It is also a strategic game since players must consider the strategies of their opponents in their decision-making process. The game explores both scenarios with a single friendly jammer and with two friendly jammers, analyzing how the number of jammers influences the game's dynamics and outcomes.

Players

- The Malicious Node: This player aims to carry out unauthorized radio transmissions within a network. The malicious node operates strategically, making decisions that maximize its ability to communicate without being disrupted by jammers.
- The Friendly Jammers: These players aim to prevent or disrupt the unauthorized communications initiated by the malicious node. The game considers two scenarios: a single friendly jammer and multiple friendly jammers operating without coordination.

Strategies

- The Malicious Node's Strategies: The strategies available to the malicious node include selecting transmission times, frequencies, and power levels that minimize the effectiveness of the jamming. The node can adapt its strategy based on the perceived actions of the friendly jammers, seeking ways to evade their interference.
- The Friendly Jammers' Strategies: For the friendly jammers, strategies involve choosing when, how, and at what power level to jam the communications of the malicious node. In the case of multiple jammers, each jammer independently

decides its strategy without coordinating with others, which introduces an element of unpredictability into the game.

Payoffs

- The Malicious Node's Payoff: The payoff for the malicious node is a function of its ability to successfully complete unauthorized transmissions despite the jamming efforts. Higher success rates in communication equate to higher payoffs, with the goal being to maximize these successful transmissions.
- The Friendly Jammers' Payoff: For the friendly jammers, the payoff is related to their ability to disrupt or prevent the unauthorized transmissions of the malicious node. Effective jamming, which leads to the failure of the malicious node's attempts to communicate, results in higher payoffs for the jammers. In scenarios with multiple jammers, the payoff also considers the added complexity the malicious node faces due to non-coordinated jamming efforts.

Outcome and Implications

The game's analysis reveals that multiple uncoordinated jammers can create a more complex and unpredictable environment for the malicious node, potentially leading to higher payoffs for the friendly jammers compared to a single jammer or coordinated efforts. This counterintuitive result underscores the potential effectiveness of deploying multiple autonomous jammers as a strategy for enhancing network security against unauthorized transmissions.

This game-theoretic model provides valuable insights into the strategic interactions between jammers and a malicious node, emphasizing the importance of strategy selection and the impact of coordination (or the lack thereof) on the success of security measures in communication networks.

Findings and Implications

One of the most striking findings of the study is that multiple uncoordinated jammers can be more effective than a single jammer or coordinated efforts. This result is counterintuitive as it suggests that a lack of coordination, often seen as a drawback, can enhance the effectiveness of jamming strategies against a malicious node. The authors attribute this outcome to the increased complexity and unpredictability that multiple uncoordinated jammers introduce, making it harder for the malicious node to anticipate and counteract the jamming signals.

This finding has profound implications for designing security measures in communication networks. It suggests that deploying multiple autonomous jammers could be a more viable strategy than focusing on coordination and control. This approach not only complicates the malicious node's counter-strategies but also introduces flexibility and scalability in security measures, essential attributes in the rapidly evolving landscape of network threats.

Conclusion

This study marks a significant step forward in applying game theory to understand and mitigate security threats in communication networks. By exploring the dynamics between friendly jammers and a malicious node, the research highlights the effectiveness of uncoordinated jamming efforts, offering a new perspective on strategic defense mechanisms. These insights contribute valuable knowledge to the meta-analysis of security problems in communication networks, emphasizing the role of game theory in developing robust security strategies. Similar studies, such as [66] and [67], further reinforce the importance of game theory in addressing security challenges within wireless sensor networks and UAV-assisted wireless communication networks, respectively. Both works underscore the utility of game-theoretic approaches in devising defense strategies against sophisticated attacks, thereby enriching our understanding of secure communication network operations.

Jamming in Underwater Sensor Networks as a Bayesian Zero-Sum Game with Position Uncertainty [15]

This paper explores a critical aspect of underwater acoustic sensor networks (UASNs) security: jamming attacks. Jamming, a deliberate attempt to interfere with the communication channel, poses a significant threat to the reliability and effectiveness of UASNs. The authors approach this problem through the lens of game theory, specifically employing a Bayesian zero-sum game framework. This approach not only addresses the adversarial nature of jamming but also incorporates the inherent uncertainty of underwater communication, particularly the uncertainty related to the positions of the nodes in the network.

Methodology

The study frames the interaction between the sensor network and a potential jammer as a zero-sum game [19] [20] [21], where one party's gain is equivalent to the other's loss. The uniqueness of their approach lies in the Bayesian model, which incorporates position uncertainty of the nodes within the game. The key challenge in underwater communication—signal attenuation based on distance and frequency—plays a crucial role in their model. By considering the positional uncertainty, the model realistically mirrors the complexities of underwater communication, setting the stage for analyzing the strategic behavior of both the network and the jammer.

The game is modeled as a two-player zero-sum game with incomplete information and structured as follows:

Type of Game

This is a Bayesian Zero-Sum Game where the sum of outcomes (payoffs) for all players across all possible outcomes is zero. In other words, one player's gain is exactly equal to the other's loss. The "Bayesian" aspect of the game incorporates uncertainty regarding certain elements of the game, such as the positions of the nodes in the network, which are not fully known to the players.

<u>Players</u>

- The Sensor Network (Maximizer): This player represents the collective efforts of the underwater sensors aiming to maintain effective communication despite interference. The network's goal is to maximize its transmission capacity, ensuring data can be relayed efficiently and accurately across the network nodes.
- The Jammer (Minimizer): The adversary seeking to disrupt the sensor network's communications. The jammer's objective is to minimize the network's transmission capacity by strategically interfering with the signals being transmitted across the network.

Strategies

The strategies for each player involve choosing actions that either maximize (for the sensor network) or minimize (for the jammer) the overall payoff, taking into account the uncertainty regarding node positions within the underwater environment.

- Sensor Network Strategies: These include varying transmission power, changing communication frequencies, or adjusting transmission protocols to mitigate interference. The network's strategies are constrained by the need to maintain effective communication despite interference and the physical limitations of underwater communication technology.
- Jammer Strategies: The jammer decides on the level of interference to apply and the targets within the network to focus its jamming efforts. Its strategies could range from constant, high-level interference across the network to more sophisticated, targeted attacks aimed at critical nodes or communication paths.







Figures 3-1 and 3-2 report the probability distribution over the BNE strategies of the players.

Payoffs

The payoffs in this game are defined by the transmission capacity of the sensor network, which the sensor network aims to maximize, and the jammer seeks to minimize.

• For the Sensor Network: The payoff is positive and increases with the network's ability to maintain high transmission capacity despite jamming. High payoff

corresponds to successful mitigation of jamming effects, ensuring robust and reliable communication across the network.

• For the Jammer: The payoff is essentially the inverse of the sensor network's payoff. A high payoff for the jammer results from successfully reducing the network's transmission capacity, effectively disrupting communication within the network.

Equilibrium and Uncertainty

The equilibrium of the game, specifically a Nash Equilibrium, occurs when both players have selected their strategies such that neither can unilaterally change their strategy to achieve a better payoff. The Bayesian aspect of the game introduces position uncertainty of the sensor nodes, adding a layer of complexity to strategy selection. This uncertainty impacts both players' strategies, as the sensor network must consider potential areas of vulnerability without precise location information, and the jammer must decide where to focus its jamming efforts without knowing exact node positions.

The outcome of this game-theoretic analysis provides insights into how underwater sensor networks can be designed and operated to mitigate the threat of jamming attacks, taking into account the complex interplay of strategies under uncertainty inherent in underwater environments.

Findings and Analysis

The analysis reveals that the equilibrium strategy for the jammer is often pure, meaning the jammer's optimal choice is to consistently interfere at a specific level, regardless of the sensor network's strategy. This insight is crucial for designing sensor networks, as it suggests that the network can anticipate a predictable interference level from intelligent jammers and adapt its communication strategy accordingly. The paper identifies specific network configurations and node positions where the impact of jamming is quantifiable and demonstrates how these positions are critical for network design and security measures. Furthermore, it highlights the dual challenge in detecting the jammer: while the network can predict the level of interference and mitigate its impact, pinpointing the jammer's location becomes significantly harder. This paper contributes to the broader meta-analysis on security issues in communication networks through a novel application of game theory. Its innovative use of a Bayesian zero-sum game to model jamming attacks in underwater sensor networks adds a unique perspective to the analysis of security challenges. By integrating position uncertainty into the game-theoretic model, the study not only addresses a specific security threat (jamming) but also enriches the game-theoretic approach to network security with a practical consideration crucial in underwater settings.

Conclusion

This study marks a pivotal enhancement in combating jamming attacks within underwater sensor networks through a Bayesian zero-sum game framework, showcasing its prowess in navigating adversarial interactions amidst uncertainty typical in underwater communications. The insights gleaned underscore the necessity of considering underwater environment's spatial dynamics and uncertainties in secure network design and operation. Augmented by similar investigations, notably [68] and [69] this paper reinforces the broad applicability and efficacy of game-theoretic methodologies in tackling security issues across diverse communication network scenarios.

A Zero-Sum Jamming Game with Incomplete Position Information in Wireless Scenarios [49]

In the dynamic landscape of wireless communications, the threat posed by jamming attacks—intentional interference aiming to disrupt legitimate communication—stands as a paramount concern. This paper ventures into this realm, presenting an insightful analysis framed within the constructs of game theory. Specifically, it addresses a scenario where a jammer and a network engage in a strategic interaction, characterized by incomplete information regarding the jammer's position, to explore the outcomes of potential jamming strategies.

Game Theory Approach

The essence of the analysis lies in modeling the interaction as a zero-sum game—a fundamental concept in game theory where one player's gain is exactly balanced by the losses of the other player(s). This approach facilitates a structured examination of the

adversarial dynamics between the jammer (the attacker) and the wireless network (the defender).

Type of Game

The game is defined as a zero-sum game with incomplete information, emphasizing the lack of complete knowledge about the jammer's location. This aspect introduces uncertainty into the strategic decisions of both players, making the game more complex and reflective of real-world scenarios.

Players

The game features two players: the jammer and the wireless network. The jammer aims to disrupt the network's communication by strategically choosing where and how to interfere. Conversely, the network seeks to maintain its communication integrity by countering the jamming attempts.

Strategies

- Jammer's Strategies: The jammer's strategies involve choosing the location and the intensity of the jamming signal. These decisions are influenced by the goal of maximizing disruption while evading detection and countermeasures.
- Network's Strategies: The network's strategies are centered on detecting the jammer and adapting its communication protocols to mitigate the interference. This includes altering transmission power, changing frequencies, or employing spatial diversity techniques.

Payoffs

The payoffs in this game are quantified by the degree of communication disruption for the jammer and the successful transmission rate for the network. For the jammer, a higher level of disruption translates to a higher payoff. For the network, maintaining higher rates of successful transmissions despite jamming represents a favorable outcome.

Analysis and Findings

The study employs a mathematical framework to analyze the strategies and payoffs, considering factors like distance-dependent signal loss and the multiplicity of access

channels with varying propagation characteristics. The analysis reveals that the game's equilibrium strategies for both players are heavily influenced by the incomplete information about the jammer's location. Specifically, it underscores the importance of strategic diversity and adaptability in the face of uncertainty.

The findings contribute valuable insights into designing more resilient wireless networks. By understanding the potential strategies of jammers and the optimal responses by networks, system designers can devise more effective countermeasures against jamming attacks.

Conclusion

The paper stands as a pivotal contribution to understanding the security of wireless networks against jamming attacks, utilizing game theory to explore the intricate dynamics between jammers and networks, especially under the veil of incomplete information. It not only broadens our comprehension of jamming in wireless environments but also sets the stage for subsequent investigations to fortify network robustness. Similar studies, such as [70] and [71], further enrich this domain by examining the interaction between smart jammers and networks and applying game-theoretical approaches to cognitive radio jamming, respectively, thereby complementing and expanding the scope of this research.

Optimal Denial-of-Service Attacks Against Status Updating [50]

This paper delves into the realm of DoS attacks targeting systems that rely on timely status updates. It stands out for its unique approach to modeling the target system through a Markov chain and an unreliable wireless communication channel. The focus is on optimizing the attack strategies to disrupt the system's ability to maintain updated status information, which is crucial for its functionality.

Game Model Description

Type of Game

The study conceptualizes the interaction between an attacker and a status updating system as a strategic game, where the attacker seeks to maximize the disruption of information flow while managing its resources.

<u>Players</u>

The game involves two main players: the adversary (attacker) aiming to launch DoS attacks and the target system attempting to update its status information reliably.

Strategies

- The adversary decides when and how intensively to attack the channel to disrupt the status updates.
- The system's strategy involves managing how it updates its status, potentially adjusting its communication protocol to mitigate the impact of these attacks.

Payoffs

The payoff for the adversary is a function of the effectiveness of the DoS attack in degrading the system's performance versus the energy or resources expended in conducting the attack. For the system, the payoff is in maintaining as accurate and timely a status update as possible under attack conditions.

Key Findings

Optimal Attack Policies: The paper demonstrates that the optimal strategy for an attacker follows a threshold-based policy. This means the attacker will initiate an attack once the system's performance or the age of information reaches a certain threshold, balancing the cost of attacking against the benefits of disrupting the system.

Performance Metrics

Two critical metrics are introduced to evaluate the system's performance under attack: the AoI and the AoII. These metrics quantify the freshness and accuracy of the status information available to the system amidst the DoS attacks.

Numerical Analysis

Through a rigorous analytical and numerical analysis, the study shows that systems measured with the AoII metric are less sensitive to jamming attacks compared to those assessed with the AoI metric. This suggests that the quality of information (correctness) might sometimes be more resilient to DoS attacks than merely the freshness of the information.

Contributions

This paper's exploration of DoS attacks in the context of status updating systems contributes to the existing literature by offering a nuanced understanding of how gametheoretic approaches can illuminate security problems in communication networks. It dovetails with similar studies, such as those conducted in [72] and [73], enriching the broader meta-analysis of security challenges within this domain. By providing a detailed examination of attack strategies and system vulnerabilities, this work particularly emphasizes the trade-offs between attack effectiveness and resource expenditure from the perspective of an attacker. Furthermore, the introduction of Age of Information (AoI) and Age of Incorrect Information (AoII) as performance metrics adds depth to our understanding of how information flow can be compromised. This, in turn, offers insights into designing more resilient systems, building on the foundational studies referenced, to address and mitigate the impacts of DoS attacks on communication networks.

Optimal Denial-of-Service Attack Energy Management Against State Estimation Over an SINR-Based Network [51]

In the evolving landscape of network security, Denial-of-Service (DoS) attacks present a significant challenge, particularly in the realm of state estimation in communication networks. The paper explores this challenge within the framework of SINR (Signal-to-Interference-plus-Noise Ratio) based networks, employing game theory to analyze the strategic interplay between attackers and network defenders. The focus is on managing the energy expended in DoS attacks to optimally disrupt state estimation processes, which are crucial for the accurate monitoring and control of networked systems.

Type of Game

The study models the conflict as a non-cooperative game, where the attacker and the network defender act as the players with opposing objectives. The game is formulated to capture the dynamic nature of the strategies employed by both sides, incorporating a continuous action space that reflects the varying levels of attack intensity and defense mechanisms.

<u>Players</u>

- Attacker: Seeks to disrupt the state estimation process by launching DoS attacks. The attacker's strategy involves choosing the optimal level of energy to expend in the attack, balancing the effectiveness of the disruption against the resources consumed.
- Defender: Aims to maintain accurate state estimation by mitigating the impact of DoS attacks. The defender's strategy involves deploying resources to counteract the interference, ensuring the integrity and reliability of the state estimation process.

Strategies

- Attacker Strategies: The choice of energy level for the DoS attack, which can range from minimal to maximal effort, depending on the attacker's resource availability and the perceived value of the disruption.
- Defender Strategies: Allocation of defensive resources to protect the state estimation process. This includes enhancing the resilience of the communication network and improving the accuracy of state estimation under attack conditions.

Payoffs

- Attacker Payoff: Defined by the level of disruption caused to the state estimation process, balanced against the cost of expending energy for the attack. The payoff is higher when the attack significantly degrades the accuracy of state estimation with minimal energy use.
- Defender Payoff: Determined by the effectiveness of the defense mechanisms in maintaining the accuracy of state estimation, balanced against the cost of defense resources. A higher payoff is achieved when state estimation remains accurate with efficient resource usage.

Analysis and Findings

The analysis reveals a complex interdependence between the attacker's and defender's strategies, where the optimal approach for each player varies with the level of resources

and the strategic responses of the opponent. The study introduces a novel approach to modeling this interaction, providing insights into the thresholds at which the cost of additional attacks outweighs the benefits and the critical points where defense strategies become cost-effective.

Implications for Network Security

This research contributes to a deeper understanding of the strategic considerations underlying DoS attacks and defense mechanisms in SINR-based networks. It underscores the importance of adaptive strategies that account for the dynamic nature of cyber threats and the continuous evolution of defense capabilities. Furthermore, the application of game theory in this context offers a valuable framework for predicting adversary behavior and informing the development of more resilient network security architectures.

Conclusion

This study stands as a significant contribution to the field of network security, offering a nuanced understanding of the energy management challenges in DoS attacks against state estimation processes. By applying game theory to this domain, the paper provides a strategic blueprint for both attackers and defenders, highlighting the critical role of resource management and strategic adaptation in safeguarding communication networks. This work complements existing research in the area, notably the findings of similar studies [74] [75], which also explore the dynamics of energy consumption and strategic defense mechanisms in the context of network security threats. Through this comprehensive approach, our study further illuminates the intricate balance between attack mitigation and energy efficiency, contributing to the development of more resilient communication networks.

3.2.2 False Data Injection and Information Security

In the digital age, information security has become a cornerstone of maintaining the integrity and reliability of communication networks across the globe. Among the plethora of cybersecurity threats, false data injection emerges as a particularly insidious technique, whereby adversaries manipulate data streams to deceive systems or users. This malicious activity not only undermines the trust in information systems but also poses significant risks to operational stability, financial integrity, and user safety. The

following section delves into the complexities of false data injection, its implications for information security, and the strategies developed to detect and counteract such attacks. Through a detailed examination, we aim to highlight the critical nature of defending against false data injection to preserve the sanctity of digital communication in our interconnected world.

Bad Data Injection Attack and Defense in Electricity Market: A Game Theory Study [28]

In the evolving landscape of smart grid technology, cybersecurity emerges as a paramount concern due to the integration of advanced cyber technologies enhancing monitoring and decision-making capabilities. However, this integration also exposes the smart grid to sophisticated cyber threats, notably bad data injection attacks. These attacks manipulate the system's data integrity, leading to potential severe technical and economical repercussions. This study delves into the dynamics of such cyber attacks within the electricity market, leveraging game theory to model the strategic interactions between attackers and defenders.

Methodology

The research conceptualizes the interaction between an attacker and a defender in the context of electricity market prices as a zero-sum game. The premise is that neither party can simultaneously attack or defend all measurements due to resource constraints. The attacker aims to alter electricity prices in their favor (either increasing or decreasing), while the defender seeks to mitigate these manipulations. The paper meticulously quantifies the impact of compromising each measurement on electricity pricing, providing a structured approach for both parties to strategize their moves effectively.



Figure 3-3 Flow of energy and data between different parts of smart grids.

Type of Game

This is a two-person zero-sum game where the sum of outcomes (payoffs) for all players for any combination of strategies equals zero. In simpler terms, one player's gain is exactly the other player's loss. This game is noncooperative, meaning each player acts independently without collaboration, aiming to maximize their own payoff while minimizing the other's. It involves both pure and mixed strategies; in cases where there is not a straightforward best move (pure strategy), players resort to mixed strategies, where they randomly choose their actions based on a probability distribution, to optimize their expected outcomes.

Players

- The Defender: Tasked with protecting the electricity market's integrity, the defender seeks to secure critical measurements from manipulation to ensure accurate state estimation and prevent financial or operational disruptions.
- The Attacker: Aims to inject false data into the system to distort state estimation and manipulate the electricity market for personal gain or to cause harm.

Strategies

• Defender's Strategies: The defender cannot guard all measurements due to resource constraints. They must strategically decide which measurements (or groups of measurements) to protect, often choosing to secure those that, if compromised, would have significant impacts. They might also deploy secure measurements in random locations to enhance detection capabilities.

• Attacker's Strategies: Similarly, the attacker cannot target all measurements. Their strategy involves selecting specific measurements whose alteration would most effectively manipulate the state estimator without detection by the control center's bad data detection (BDD) systems.

Payoffs

The payoff is based on the success of the strategies employed by each player:

- Defender's Payoff: Successfully protecting critical measurements and detecting false data injections maintains the integrity of the electricity market, minimizing financial and operational disruptions. The better the defender is at securing the network and identifying attacks, the higher their payoff.
- Attacker's Payoff: Successfully injecting false data without detection allows the attacker to manipulate the market as desired, potentially gaining financially or causing harm. The payoff is higher for undetected, impactful manipulations.

Game Dynamics and Solution

The interaction between defender and attacker strategies can be described using matrix games and solved using linear programming to find the mixed-strategy equilibrium. The equilibrium in mixed strategies ensures that, on average, neither player can improve their outcome by unilaterally changing their strategy mix, given the strategy mix of the opponent.

- <u>Mixed Strategies:</u> These involve players randomizing their actions according to probability distributions, enabling a strategic balance when a pure strategy does not offer a clear advantage.
- Saddle-Point Equilibrium: The point in the game where the selected strategies by both players result in stabilized payoffs, where neither player can benefit by changing their strategy while the other keeps theirs unchanged.
- Linear Programming (LP): Converts the game into an LP problem to compute the optimal mixed strategies for both players. This approach identifies the minimum expected loss for the defender and the maximum expected gain for

the attacker under the equilibrium condition, providing a balanced outcome based on the mixed strategies.

Findings

The study employs the PJM 5-Bus test system for simulation, illustrating the feasibility and implications of the proposed game-theoretic model. Key insights from the simulations underscore:

- Strategic Interaction: The attacker and defender dynamically adjust their strategies based on the other's actions, showcasing the intricate balance of attack and defense strategies in safeguarding electricity market integrity.
- Effectiveness of Game Theory: The game-theoretic approach effectively captures the complexity of interactions in cybersecurity scenarios within electricity markets. It identifies the proportional times and measurements that attackers and defenders are likely to target, offering nuanced insights into optimizing defense mechanisms.
- Economic and Technical Impact: By simulating various attack and defense scenarios, the study highlights the potential economic and technical vulnerabilities in the electricity market arising from bad data injection attacks. It also underscores the effectiveness of strategic defense measures in mitigating these impacts.

Conclusion

This research is pivotal in understanding and countering bad data injection attacks in smart grids through a game-theoretic lens, aligning closely with the frameworks and findings of similar studies [76][77]. It offers a comprehensive framework for analyzing the strategic behavior of attackers and defenders, shedding light on the economic and technical vulnerabilities of electricity markets to cyber threats. In conjunction with the methodologies and results presented in [76] and [77], this study contributes significantly to enhancing the security posture of smart grids. Furthermore, it paves the way for further research in applying game theory to tackle cybersecurity challenges in communication networks, extending the dialogue initiated by these seminal works.

Data Injection in a Vehicular Network Framed Within a Game Theoretic Analysis [23]

This paper leverages game theory to explore and mitigate data injection threats, offering insights into strategic defense mechanisms in a vehicular ad hoc network (VANET). This summary encapsulates the key findings and methodologies of the study, highlighting its contribution to the field of network security.

Introduction

As vehicles increasingly become interconnected, the vulnerability of vehicular networks to cyber-attacks, particularly data injection threats, has become a significant concern. The intricate nature of these networks, facilitating communication between vehicles and roadside infrastructures, necessitates robust security solutions. The authors address this challenge through a game-theoretic lens, aiming to understand and counteract the strategies of potential attackers.



Figure 3-4 RSU and SV interaction in a VANET

Methodological Approach

Employing the mathematical rigor of game theory, the authors construct a model that simulates the strategic interactions between attackers and defenders within a vehicular network. The study utilizes the *nashpy* [24] Python package for the computation of Nash Equilibria, employing the support enumeration algorithm to identify equilibrium points where neither party (smart vehicles nor roadside units) can unilaterally improve their payoff by changing their strategy.

Strategic Framework

The game-theoretic model devised by the authors discretizes the strategic spaces for both smart vehicles (SVs) and roadside units (RSUs), enabling a detailed exploration
of potential actions and their outcomes. This approach allows the study to map out the conditions under which the network can achieve an optimal balance between data exchange efficiency and security against data injection attacks.

Type of the Game

- Static Game of Complete Information: Initially, the interaction is modeled as a static game where all players (SVs and RSUs) have complete information about each other's strategies and payoffs. This model aims to find an equilibrium solution where SVs submit data at a rate that is sustainable for the network while still fulfilling their need for timely and accurate information exchange.
- Dynamic Repeated Game: The game evolves into a dynamic, repeated scenario reflecting the continuous and changing interactions between SVs and RSUs. This model considers the variable nature of VANETs, including SVs' movement speeds, directions, and the short but critical connections they establish. The repeated nature of the game accounts for the ongoing decision-making process of data submission and bandwidth allocation, with past interactions potentially influencing future strategies.

Players

- Smart Vehicles (SVs): SVs aim to submit as much data as necessary to maintain an up-to-date representation of their environment. Their challenge is to balance the need for frequent, considerable data transmission against the limitations of the network's bandwidth and the potential for creating congestion or enabling security vulnerabilities.
- Road-Side Units (RSUs): RSUs serve as gateways for data exchange in the VANET and play a crucial role in managing data injection from SVs. They make strategic decisions on resource allocation to prevent network saturation and misuse, including DoS attacks by malicious SVs.

Strategies

• SVs' Strategy: Choose the rate of data injection. This decision involves determining how much data to submit to maintain effective communication without overwhelming the network or appearing malicious.

• RSUs' Strategy: Independently set a target for the SVs' data injection rate. RSUs aim to manage the network's bandwidth effectively, allowing for efficient data exchange without risking saturation or facilitating attacks.

Payoffs

- SV Payoffs: SVs seek to maximize their ability to transmit essential data for safety and decision-making purposes. Their payoff is a balance between effective communication and the risk of restricted access due to perceived misuse.
- RSU Payoffs: RSUs aim to maximize network efficiency and security. Their payoff involves maintaining optimal bandwidth usage and preventing malicious behavior, ensuring that SVs can share critical data without compromising the network's integrity.

Equilibrium

- In the static game, an equilibrium is achieved when SVs submit data at a rate that matches the RSUs' target, allowing for some flexibility in data injection but preventing network abuse.
- In the dynamic, repeated game, equilibrium involves adapting strategies over time, possibly leading to implicit agreements or collaboration between SVs and RSUs to optimize network usage and safety continually. This adaptive approach allows for more nuanced cooperation and response to changing network conditions and threats.



Figure 3-5 Payoff Surfaces

Findings

The research uncovers a unique Nash Equilibrium, indicating a set of strategies for SVs and RSUs that maximizes network security and data exchange efficiency. One key insight is the delineation of threshold values for the RSU's traffic targets, beyond which the network becomes particularly vulnerable to attacks. This equilibrium suggests a strategic template for network administrators to optimize security protocols and safeguard against malicious data injections.

Discussion

The paper's findings illuminate the intricate dynamics of security in vehicular networks, showcasing the utility of game theory in crafting defense strategies. The study not only enhances our understanding of the strategic considerations underlying network security but also proposes a novel approach to mitigating cyber threats through equilibrium-based planning.

Implications for Network Security

The application of game theory as demonstrated in this study offers several advantages for the field of network security, including predictive modeling of attacker behavior, dynamic adaptation to evolving threats, and efficient allocation of defensive resources. By providing a framework for analyzing the strategic interplay between network entities, the research contributes significantly to the development of more resilient security mechanisms in vehicular networks and beyond.

Conclusion

This study marks a substantial advancement in integrating game theory into network security paradigms, particularly within the realm of vehicular networks. Drawing from the strategic insights offered by our investigation, alongside similar explorations noted in similar studies [78] [79], it becomes evident that a deeper understanding of the strategic underpinnings of cybersecurity is crucial for devising more effective defense mechanisms. The emphasis on strategic planning and the pursuit of equilibrium-based tactics, as highlighted in our findings and corroborated by these parallel studies [78] [79], underscore the pivotal role that such approaches play in mitigating cyber threats. As vehicular networks undergo further development and sophistication, the principles and methodologies outlined in this paper, along with the insights garnered from studies

[78] [79], are poised to significantly influence the formulation of future security strategies.

Analysis of Strategic Security Through Game Theory for Mobile Social Networks [58]

This paper focuses on addressing security concerns in mobile social networks through game theory. The authors aim to identify malicious nodes within these networks, proposing a Bayesian game framework that includes a broader set of actions for players, notably the option to engage in packet exchanging, malicious activity, and its prevention.

Game Theory Framework

Type of Game

Bayesian Game, where the critical parameter is the likelihood of an agent being malicious.

Players

- Server: Represents legitimate nodes within the network, which can decide to transmit packets, monitor the network for malicious nodes, or remain inactive.
- Client: Represents nodes whose intentions are unknown and can be benign or malicious. This player can forward a received packet (benignly), ignore it, or corrupt it (maliciously).

Strategies

- Server: Can choose to do nothing, send packets, or engage in surveillance.
- Client: Depending on its nature (benign or malicious), can decide to forward, ignore, or damage (corrupt) the packets.

Payoffs

The payoffs are determined by the actions of both players, with specific outcomes for different combinations of strategies. The game aims to find Nash Equilibria in both pure and mixed strategies to understand the optimal strategies under different conditions (e.g., when the client is likely benign or malicious).

Key Findings

The analysis reveals various Nash Equilibria, indicating that under certain conditions, the presence of malicious nodes can be tolerated to maintain a trade-off between security measures and network performance. The game's outcomes suggest that the best strategy for the network (server) might not always involve direct identification and counteraction against malicious nodes. Instead, it could involve forcing malicious nodes to adopt less harmful strategies or ensuring that the cost of malicious activities outweighs their benefits.

The authors propose that the effectiveness of network security strategies can be enhanced by understanding the strategic behaviors of nodes within mobile social networks. This approach allows for a low-cost, distributed management of network security, providing theoretical performance bounds and insights into designing efficient defense mechanisms.

Implications for Network Security

This work has significant implications for designing network security mechanisms in mobile social networks. By employing game theory, the research provides a structured way to analyze the strategic interactions between different types of nodes within the network, offering a method to anticipate and mitigate potential security threats in a distributed and cost-effective manner. The paper suggests that understanding the probabilistic nature of malicious behavior and incorporating strategic decision-making can lead to more resilient mobile social networks.

In summary, this paper introduces a novel approach to analyzing and enhancing security in mobile social networks through the application of game theory. It emphasizes the significance of strategic decision-making in addressing network security challenges. This study builds upon and extends the insights provided by similar research in the field, notably those presented in references [80] and [81]. These studies collectively underscore the growing relevance of game-theoretic models in comprehending and mitigating security threats within mobile social networks. Our contribution further enriches the discourse by offering unique perspectives on the application of these models to enhance network security.

3.2.3 Eavesdropping and Information Security

In the complex world of communication networks, eavesdropping attacks stand out as a critical security issue. These attacks, where unauthorized individuals secretly intercept and listen to private communications, represent a significant challenge within the broader context of digital communication security. Eavesdropping can occur in various forms, including the interception of emails, instant messages, and even phone calls, highlighting the diverse vulnerabilities present in today's communication infrastructures. Against this backdrop, there is a pressing need for effective defense mechanisms.

This need has spurred the adoption of game theory as an innovative analytical tool in the exploration of security problems within communication networks. By treating security challenges as strategic games between attackers and defenders, this part aims to uncover new insights and strategies for combating eavesdropping attacks. Focusing on eavesdropping as a key issue, the study ventures into the game-theoretical approaches to network security, seeking to identify robust solutions that can protect against such intrusions.

Thus, this research not only sheds light on the intricacies of securing communication networks against eavesdropping but also proposes a forward-looking perspective on developing more secure and resilient systems through the application of game theory. This approach promises not only to enhance our understanding of network vulnerabilities but also to guide the formulation of more effective security policies and technologies.

User Cooperation Analysis Under Eavesdropping Attack: A Game Theory Perspective [25]

In the dynamic landscape of communication networks, ensuring secure transmissions against eavesdropping attacks is paramount. The paper delves into the realm of physical layer security, applying game theory to scrutinize user cooperation behavior when faced with eavesdropping threats. This innovative approach uncovers the limitations of conventional cooperation schemes and proposes an adaptive strategy that aligns with the incentives of network users, marking a significant step towards enhancing network secrecy.

Analysis Framework

The study employs a game-theoretic framework to model the interactions between users and eavesdroppers. Traditional cooperation methods, such as relaying information through intermediate nodes, are evaluated under the assumption that eavesdroppers may have a superior channel condition. The analysis reveals a counterintuitive outcome: conventional cooperation could actually worsen the secrecy performance compared to direct transmissions. This finding is pivotal as it challenges the prevailing notion that user cooperation invariably benefits network security.

Here is a breakdown of the game:

Type of Game

The game explored is a strategic non-cooperative game with elements of cooperative game theory embedded within the adaptive scheme proposed. It intricately combines aspects of both Nash equilibrium analysis (characteristic of non-cooperative games) and Stackelberg games, especially when introducing the adaptive cooperation scheme with a punishment mechanism.

<u>Players</u>

There are two main types of players in this game:

- Users: These are the individuals or nodes within the network that wish to communicate securely. They are interested in maximizing the secrecy of their transmissions.
- Eavesdropper: This player aims to intercept and decode the transmissions between users. The eavesdropper seeks to maximize its ability to overhear the communications, thus minimizing the secrecy performance of the network.

Strategies

The strategies available to the players are centered around whether or not to cooperate and how to do so:

Users' Strategies:

- Cooperate: Participate in the cooperative scheme by relaying messages for other users, potentially improving the overall secrecy of the network.
- Not Cooperate: Directly transmit messages to the intended recipient without the involvement of intermediate nodes, based on the assessment that cooperation does not offer additional utility under certain conditions.

Eavesdropper's Strategy:

• Channel Selection: Choose the channel conditions or positions relative to the users that maximize the ability to eavesdrop on the transmissions.

Payoffs

The payoffs in this game are derived from the secrecy performance of the network, which is influenced by the strategies of both the users and the eavesdropper:

Users' Payoffs: The utility for users is related to the secrecy performance of their transmissions. High secrecy performance results in higher payoffs, reflecting the successful concealment of communication from the eavesdropper.

In the adaptive cooperation scheme the payoff for users also includes the additional utility gained from participating in cooperation, even when the eavesdropper has superior channel conditions.

Eavesdropper's Payoff: The payoff for the eavesdropper is inversely related to the network's secrecy performance. Greater ability to intercept and decode transmissions leads to higher payoffs, indicating a lower level of network security.

Game Theoretic Insights

The paper's key insight is that conventional cooperation schemes can actually deteriorate the secrecy performance under specific conditions, specifically when the eavesdropper has better channel conditions compared to the destination. To motivate user cooperation and improve secrecy performance, the authors propose an adaptive cooperation scheme. This scheme is designed to be a Nash equilibrium, where users are incentivized to cooperate because doing so increases their payoffs compared to not cooperating, even in the presence of a strong eavesdropper. The study further introduces

a Stackelberg game with a punishment mechanism to ensure that mutual cooperation becomes the unique Nash equilibrium, meaning that it is in the best interest of all users to participate in the cooperation game to maximize their payoffs.

Key Findings

A central discovery is the identification of conditions under which user cooperation ceases to be beneficial for enhancing security. Specifically, when the eavesdropper enjoys a better channel condition than the intended destination, users lack the motivation to participate in cooperative strategies, given that such participation does not yield additional utility. This scenario underscores a critical gap in the traditional approach to securing communication networks.

To address this challenge, the authors propose an adaptive cooperation scheme designed to improve secrecy outcomes, irrespective of the eavesdropper's channel advantage. This scheme encourages mutual cooperation among users by ensuring that cooperation becomes a Nash equilibrium, a stable state in which no participant can benefit by unilaterally changing their strategy. Furthermore, by integrating a Stackelberg game with a punishment mechanism, the research ensures that mutual cooperation emerges as the unique Nash equilibrium, thereby motivating user participation in the cooperation game.

Implications for Communication Networks

The implications of this research for the security of communication networks are profound. By illustrating the conditions under which traditional cooperation strategies fail and proposing an adaptive scheme that aligns with user incentives, Niu et al. offer a pathway to enhance network security in the face of eavesdropping attacks. This contribution is particularly relevant in the era of ubiquitous wireless communications, where the risk of eavesdropping is ever-present.

Moreover, the application of game theory to the analysis of security problems opens new avenues for research. It provides a rigorous framework for evaluating the effectiveness of security strategies, taking into account the strategic behavior of network participants. This approach not only enriches the theoretical understanding of network security but also offers practical insights for designing more robust communication systems.

Conclusion

This study marks a considerable leap forward in the realm of network security research. Incorporating a rigorous game-theoretic analysis, this paper questions the status quo, exposes the inadequacies of classical cooperation approaches, and proposes a novel adaptive strategy poised to enhance network confidentiality. In doing so, it not only enriches the theoretical landscape of network security but also bears significant practical ramifications for the crafting and management of secure communication frameworks. Moreover, by echoing and building upon the insights offered in similar studies [82] [83], this work stands as an invaluable component of the broader discourse on dissecting and addressing security challenges within communication networks via a game-theoretical lens.

A Game Theoretic Approach to Eavesdropper Cooperation in MISO Wireless Networks [26]

The paper addresses a pivotal concern in the realm of information theoretic security, particularly focusing on the challenge posed by eavesdroppers in wireless networks. With an increasing volume of private data being transmitted over wireless channels, which inherently exhibit broadcasting characteristics, the study emphasizes the critical need for a robust security framework. Traditional approaches in secrecy analysis have often simplified the scenario by considering the presence of a singular eavesdropper or disregarding potential cooperation among multiple eavesdroppers. This paper diverges from conventional studies by exploring the dynamics of eavesdropper cooperation within Multiple-Input Single-Output (MISO) wireless communication systems, employing a game theoretic analysis to shed light on the strategic interactions and countermeasures relevant to securing secret information transmission.

In this study the game theoretic framework is designed to analyze the interactions between a transmitter (Alice) and multiple potential eavesdroppers in a wireless communication environment. The game aims to explore the strategic behaviors of both the transmitter and the eavesdroppers, especially focusing on the conditions under which eavesdroppers may choose to cooperate or act independently. Here's a breakdown of the game:

Type of Game

The study models the interaction as a non-cooperative game. Specifically, it is described as a "one-shot eavesdropper cooperation game" for the analysis of immediate decisions without future interactions and an "infinitely repeated game" for scenarios involving ongoing interactions. The analysis aims to understand how eavesdroppers decide whether to cooperate or act independently based on the potential outcomes of their actions.

<u>Players</u>

- Transmitter (Alice): Alice represents the sender of secret information, who is trying to communicate securely over a wireless network. She is interested in preventing eavesdroppers from intercepting the transmitted data.
- Eavesdroppers: Individuals or entities attempting to intercept the communication between Alice and her intended recipient. Eavesdroppers can decide whether to work alone or cooperate with other eavesdroppers to improve their chances of successfully intercepting the data.

Strategies

Alice's Strategies:

- Non-Preventive Strategy: Alice transmits without any specific countermeasures against eavesdropping.
- Preventive Strategy: Alice detects eavesdropping attempts and responds by degrading the eavesdropping channel. This can involve transmitting artificial noise or employing other techniques to reduce the intelligibility of the intercepted data.

Eavesdroppers' Strategies:

• Non-Cooperation: Eavesdroppers work independently, each attempting to intercept the data without collaborating with others.

• Cooperation: Eavesdroppers form a coalition, sharing resources and information to improve their collective probability of successfully intercepting the communication.

Payoffs

- Alice's Payoff: Alice's payoff is related to the level of security she can maintain for her communication. The higher the security, the higher her payoff. Her payoff decreases as the eavesdroppers' ability to intercept and understand the transmitted data increases.
- Eavesdroppers' Payoff: The eavesdroppers' payoff is determined by their success in intercepting and decoding the transmitted data. Successful eavesdropping increases their payoff, while failed attempts or degraded interception quality due to Alice's preventive strategies decrease their payoff.

Outcome: Nash Equilibrium and Cooperation Conditions

The analysis in the paper reveals that non-cooperation among eavesdroppers can be a Nash Equilibrium in a one-shot game, implying that, given the preventive strategies of Alice, eavesdroppers might find it in their best interest to act independently rather than cooperate. However, in infinitely repeated games, conditions might arise under which eavesdroppers' cooperation could emerge as a more beneficial strategy for them, challenging the security measures implemented by Alice.

Key Insights and Findings:

Nash Equilibrium in Eavesdropping: The paper derives a significant conclusion that non-cooperation among eavesdroppers represents a Nash Equilibrium in a one-shot game scenario. This outcome indicates that in a single interaction, individual eavesdroppers are better off acting independently rather than cooperating, due to the countermeasures employed by the transmitter.

Conditions for Eavesdropper Cooperation: The analysis extends to infinitely repeated games, revealing conditions under which cooperation among eavesdroppers could be sustainable and potentially more harmful to the security of communications. These conditions highlight the delicate balance between the defensive strategies of the transmitter and the adaptive behaviors of eavesdroppers in a dynamic environment.

Implications for Wireless Network Security: The findings underscore the complexity of securing wireless networks against eavesdropping. They demonstrate that while certain transmitter strategies can effectively discourage eavesdropper cooperation in specific instances, a comprehensive security framework must consider the potential for adaptive and cooperative behaviors among eavesdroppers over extended periods.

Conclusion:

This study contributes to the body of knowledge on information theoretic security by providing a nuanced understanding of eavesdropper cooperation in MISO wireless networks through a game theoretic lens. It challenges the traditional assumptions of non-cooperative eavesdroppers and elucidates the conditions under which eavesdropper cooperation becomes a viable strategy. In this regard, our findings are in line with similar studies that have explored the application of game theory to security issues in communication networks. Notably, the works of [84] and [85] offer complementary perspectives on the use of game theory to address security challenges, including cooperative jamming and dynamic defense mechanisms in wireless and network security settings. By integrating insights from these studies, our analysis underscores the importance of developing dynamic and adaptive security mechanisms that can anticipate and counteract evolving threats in wireless communication systems. The insights derived from our analysis, along with those of [84] and [85], are critical for researchers and practitioners alike in devising more effective strategies to safeguard private information transmission in an increasingly interconnected and wireless world. This collective body of work reinforces the value of game theoretic approaches in enhancing the security and resilience of communication networks against sophisticated threats.

A Secure Communication Game with a Relay Helping the Eavesdropper [29]

This study offers a novel perspective on enhancing network security through game theory, specifically addressing a scenario where a relay assists an eavesdropper, thus posing a unique challenge to secure communication.

Theoretical Framework

The paper examines a four-terminal Gaussian network composed of a source, a destination, an eavesdropper, and a uniquely positioned jammer relay. Unlike traditional models where relays support the source in overcoming eavesdropping, this setup features a relay that aids the eavesdropper. This contrarian approach introduces a dynamic interplay between maximizing and minimizing achievable secrecy rates, encapsulated through a two-player zero-sum game framework. The essence of the game lies in the strategic dissemination of information (by the source) and interference (by the jammer relay) to manipulate secrecy rates.

Methodology

Employing a rigorous mathematical model, the authors assume Gaussian strategies for both the source and the jammer relay's transmissions. The complexity of the interactions necessitates the formulation of the problem as a continuous game, where the payoff is the achieved secrecy rate. A pivotal aspect of this analysis is the determination of Nash Equilibrium through mixed strategies, illustrating the probabilistic nature of optimal play for both parties.

Type of Game

The game is a two-player zero-sum continuous game. In zero-sum games, the gain (or loss) of one player is exactly balanced by the losses (or gains) of the other player(s). A continuous game allows the players to choose their strategies from a continuous spectrum of options.

Players

The Source: The source aims to transmit data securely to its destination, endeavoring to maximize the secrecy rate, which is the rate at which data can be securely transmitted to the destination without the eavesdropper being able to intercept it effectively.

The Jammer Relay: Contrary to conventional roles where a relay aids the source, this relay assists the eavesdropper by attempting to minimize the achievable secrecy rate. Its goal is to disrupt the communication between the source and its destination as effectively as possible.

Strategies

Source Strategy: The source can choose how much information to send and how to encode it, considering the potential for eavesdropping. Gaussian strategies imply that the source varies its transmission power and encoding methods based on a probabilistic distribution, seeking to optimize the secrecy rate.

Jammer Relay Strategy: Similarly, the jammer relay employs Gaussian strategies, adjusting its jamming power and interference patterns to maximize the disruption to the source's transmission. The relay's strategy is also based on a probabilistic distribution, aiming to minimize the secrecy rate achievable by the source.

Payoffs

For the Source: The payoff is the achievable secrecy rate, the rate at which it can transmit data securely to the destination. The source aims to maximize this rate, thus ensuring that the communication is as private as possible.

For the Jammer Relay: Since this is a zero-sum game, the jammer relay's payoff is essentially the negative of the source's payoff. It aims to minimize the secrecy rate, effectively reducing the security of the communication between the source and the destination.

Nash Equilibrium

The Nash Equilibrium in this game generally involves mixed strategies, where both the source and the jammer relay select their strategies based on optimal cumulative distribution functions. At equilibrium, neither player can unilaterally change their strategy to improve their payoff, indicating the balance between maximizing and minimizing the secrecy rate.

Findings

The crux of the findings is the identification of optimal cumulative distribution functions for the source and the jammer relay, which dictate the probability distribution of their respective actions to achieve equilibrium secrecy rates. This equilibrium represents a state where neither the source nor the jammer can unilaterally improve their position, highlighting the nuanced balance between enhancing and compromising communication security.

Implications for Network Security

This study illuminates the intricate balance between offensive and defensive strategies in network security. The application of game theory unveils the strategic depth behind seemingly straightforward security measures, advocating for a more sophisticated analysis of potential threats and countermeasures in communication networks. Furthermore, it underscores the potential of adversarial relays in disrupting secure communication, prompting a reevaluation of network design and security protocols to mitigate such unconventional threats.

Conclusion

This research marks a seminal contribution to understanding security in communication networks from a game-theoretical perspective, significantly broadening the scope of security analysis. By including adversarial internal network elements and providing a strategic framework to anticipate and neutralize threats, it highlights the multifaceted nature of network security. Similar studies, such as those found in [86] and [87], further underscore the invaluable insights game theory offers in devising robust defensive strategies, reinforcing the comprehensive approach to tackling security challenges in communication networks.

3.3 SECURITY PROBLEMS ADDRESSING AGE OF INFORMATION (AOI)

The Age of Information (AoI) constitutes a pivotal concept in the domain of communication networks, serving as a metric to assess the freshness or timeliness of the information being delivered to recipients. This concept shifts the focus from traditional metrics such as throughput and latency to the age or staleness of the information from the moment it was generated until it reaches the end-user. Essentially, AoI measures the time elapsed since the last received packet of data was generated, offering a unique perspective on network performance by emphasizing the relevance of delivering up-to-date information in various applications, including real-time monitoring and control systems. By prioritizing the freshness of information, AoI provides an innovative framework to evaluate and enhance communication protocols and network designs, ensuring that the data utilized for decision-making processes is as

recent as possible, thereby enhancing the efficiency and responsiveness of systems reliant on timely data [48].

3.3.1 Jamming Attacks and Network Security

A Dynamic Jamming Game for Real-Time Status Updates [52]

This paper investigates the interplay between a status updating system and a malicious jammer within the context of communication networks. It is a pioneering work that merges the concepts of the Age of Information (AoI) with game theory to address security concerns in real-time status update systems. This study is vital for understanding how to maintain information freshness in the presence of adversarial threats, which is critical for various applications, from sensor networks to autonomous driving.

Game-Theoretical Framework

The study models the interaction between the updating system and the jammer as a dynamic game, which is a strategic situation where the participants (players) make decisions sequentially, and their payoffs depend on the combination of choices made.

Type of Game

The game is dynamic and incomplete information, where the jammer does not know precisely when new status updates are generated but has knowledge about the system's updating frequency.

Players:

The two main players in this game are <u>the status updating system</u> and the jammer. The updating system aims to send fresh updates to a receiver, while the jammer's goal is to disrupt this process by blocking the communication channel.

Strategies:

• The updating system's strategy involves deciding the timing for sending updates to minimize the average AoI, considering the jamming threats.

• The jammer's strategy is about choosing when and how intensely to jam the channel to maximize the disruption of the update process, constrained by its energy resources.

Payoffs:

- The updating system's payoff is inversely related to the AoI, aiming for the lowest possible AoI despite jamming.
- The jammer's payoff is a function of the increase in AoI it causes, balanced against the cost of jamming (e.g., energy consumption).

Analysis and Results

The interaction between the jammer and the updating system is analyzed through a game-theoretic lens, revealing the strategic nature of their decisions. The study demonstrates the existence of a Nash Equilibrium where both players adopt strategies that, once settled, leave no player with the incentive to deviate. At this equilibrium, the updating system optimizes the timing of its updates to minimize AoI, considering the jamming activities, while the jammer strategically allocates its jamming efforts to maximize disruption without overspending its resources.

The results highlight the delicate balance between ensuring timely updates and mitigating the impact of jamming. They suggest that the updating system can adopt unpredictability in its update timings as a countermeasure against jamming, which aligns with broader security practices of unpredictability in defense mechanisms.

Conclusion

This paper provides a novel insight into securing communication networks against jamming attacks using a game-theoretic approach, highlighting the significance of strategic planning in safeguarding information freshness amidst security threats. The dynamic jamming game model proposed lays the groundwork for future research on optimizing communication strategies under adversarial conditions, significantly enriching the discourse on Age of Information (AoI) and security. This contribution is bolstered by similar studies [88][89], which explore game-theoretic learning antijamming approaches and the analysis of on-off jamming scenarios, respectively. Together, these works underscore the theoretical and practical benefits of applying game theory to enhance communication network resilience.

Maintaining Information Freshness Under Jamming [53]

This paper addresses the challenge of maintaining the freshness of information (referred to as the Age of Information, AoI) in unmanned aerial vehicle (UAV) communications with ground control stations (GCS) in the presence of hostile interference, such as jamming. It employs a game-theoretic framework to model the interactions between a UAV transmitter and an adversarial jammer, aiming to understand and devise strategies that can mitigate the impact of jamming on the communication system's ability to maintain up-to-date information.

Game-Theoretical Model

Type of Game

The game is analyzed through both Nash and Stackelberg equilibrium concepts, highlighting the strategic interactions between the UAV transmitter and the jammer. The paper explores these equilibria under different conditions, including the presence of background noise and the impact of the transmitter's packet updating rate on the strategies.

<u>Players</u>

- The UAV transmitter, which aims to update its information at the GCS while minimizing the cost associated with its transmission power.
- The adversarial jammer, which seeks to disrupt the UAV's communication by maximizing its own utility, which is modeled as the difference between the AoI impact and the cost of jamming.

Strategies

- The UAV transmitter's strategy is characterized by its choice of transmission power level, which affects the rate at which information updates are sent to the GCS.
- The jammer's strategy is defined by its jamming power level, which impacts the interference experienced by the UAV's communication channel.

Payoffs

- The UAV transmitter's payoff is influenced by the AoI at the GCS and the cost of transmitting at a particular power level.
- The jammer's payoff is determined by the effectiveness of its jamming (as measured by the AoI) and the cost of jamming.

Summary of Key Findings

The paper demonstrates that the introduction of background noise and the consideration of the transmitter's packet updating rate significantly enrich the game's strategic landscape. Specifically, it shows that:

- <u>Nash Equilibria</u>: In the presence of background noise, Nash equilibrium strategies are derived, revealing how these strategies depend on the transmitter's packet updating rate and the sensitivity of the transmitter to network parameters.
- <u>Stackelberg Equilibria:</u> The paper identifies conditions under which Stackelberg equilibria arise, emphasizing the hierarchical relationship between the transmitter and the jammer when the former acts as the leader. This is a notable deviation from scenarios without background noise, where such an equilibrium may not exist.

The analysis of Nash and Stackelberg equilibria reveals that multiple equilibrium strategies can arise, depending on the network parameters and the presence of background noise. The paper further discusses how these equilibria affect the probability of mission success by maintaining information freshness under jamming conditions.

In conclusion, this work significantly contributes to the understanding of security problems in communication networks approached through game theory. It not only extends existing models by incorporating the effects of background noise and transmission rate sensitivity but also provides comprehensive insights into the strategic considerations necessary to mitigate the impact of jamming on information freshness. Our findings resonate with similar studies in the field, such as the work by Costa et al. [90], which delves into adaptive coding strategies to combat jamming, and the study on

timely and covert communications under eavesdropping and jamming effects [91]. These studies, akin to ours, highlight the evolving complexity of ensuring information freshness in the face of adversarial actions in networked communication systems. Together, they underscore the critical need for innovative approaches that combine game theory with practical coding and transmission strategies to safeguard against sophisticated jamming and eavesdropping tactics.

Game theoretic analysis of an adversarial status updating system [61]

This paper investigates the equilibrium points in a status updating system challenged by an adversarial entity that jams the updates in the downlink. Focusing on both diversity and non-diversity system models, the study reveals that while Nash equilibria might be elusive in some settings, Stackelberg equilibria can emerge, particularly when the base station's scheduling algorithm acts as the leader against the adversary's actions.

System Models and Problem Formulation

The investigation covers two primary settings: one without diversity, where the base station schedules updates to users without frequency diversity, and another with diversity, introducing multiple sub-carriers for update transmissions. In both models, the adversary aims to jam communications, constrained by the ability to block only a certain proportion (α) of the total communication window.

Game Theoretic Framework

Type of Game

The game is a strategic interaction with elements of both Nash and Stackelberg equilibria, depending on the model (with or without diversity). In the non-diversity setting, the game leans towards a Stackelberg competition, where the base station (leader) makes a move (scheduling decision) that the adversary (follower) responds to. In the diversity setting, the game is framed to find a Nash equilibrium, where both players' strategies are in equilibrium; neither has an incentive to deviate given the strategy of the other.

Players

• Base Station: Acts as a scheduler, deciding how to allocate time slots for updates to users in the network.

• Adversary: Seeks to disrupt the communication by blocking or jamming updates during selected time slots.

Strategies

- Base Station's Strategy: Involves deciding on a scheduling algorithm to determine which user receives updates at each time slot. In the diversity model, this also includes choosing sub-carriers for transmitting updates.
- Adversary's Strategy: Consists of choosing time slots and, in the case of the diversity model, sub-carriers to block, with the objective of maximizing the age of information and thus degrading the quality of the status updates.

Payoffs

- Base Station's Payoff: The base station aims to minimize the average age of information across all users, thus ensuring that the status information remains as fresh as possible. The payoff is essentially the negative of the average age of information.
- Adversary's Payoff: The adversary's goal is to maximize the average age of information, thereby reducing the freshness of the status updates. The payoff is the increase in the average age of information caused by its jamming actions.

The analysis frames the interaction between the base station and the adversary as a game, exploring the existence and characterization of Nash and Stackelberg equilibria. The findings indicate that:

- Without Diversity: A Nash equilibrium does not exist, but a Stackelberg equilibrium is achievable with the base station leading. The optimal strategies involve the base station scheduling updates in a manner that minimizes the overall system's age of information (AoI), considering the adversary's blocking actions.
- With Diversity: The introduction of sub-carriers allows for a Nash equilibrium where both the base station and the adversary adopt strategies that involve scheduling and blocking actions, respectively, that optimize the AoI under the given constraints.

Key Results and Implications

The research delineates the conditions under which equilibria exist and the strategies that lead to these equilibria. The Stackelberg equilibrium in the non-diversity model showcases the preemptive advantage of the base station's scheduling strategy over the adversary's blocking actions. Conversely, the existence of a Nash equilibrium in the diversity model highlights the possibility of reaching a state of mutual optimal strategies, balancing the freshness of information against adversarial interferences.

Conclusion

This paper contributes to the understanding of security challenges in communication networks from a game-theoretic perspective. By examining the strategic interactions between a base station and an adversarial jammer in the context of Age of Information (AoI), it lays the groundwork for future research on optimizing communication strategies in the presence of adversarial threats. The analysis presented complements similar studies in the field, such as the investigation of deception in network security through game theory [92], and the exploration of multistage attack graph security games [93]. These studies collectively enhance our comprehension of how game-theoretic concepts can be applied to address and mitigate security risks in communication networks. Our findings hold significant promise for designing robust wireless networks that are resilient to adversarial threats, especially in critical real-time applications. Together, these contributions mark a step forward in the development of strategic defense mechanisms against sophisticated cyber-attacks, highlighting the pivotal role of game theory in securing modern communication infrastructures.

Age of information of a power constrained scheduler in the presence of a power constrained adversary [62]

This paper investigates the age of information (AoI) in a communication network with a power-constrained scheduler (base station) and adversary amidst N users and Ns communication channels. This setup models real-world scenarios where maintaining fresh information is critical, and it is complicated by adversarial actions such as jamming or blocking communications.

Game-Theoretical Model

Type of Game

The study approaches the problem as a game between the base station (scheduler) and an adversary, both subject to power constraints.

<u>Players</u>

- Base Station (Scheduler): Aims to minimize the AoI by scheduling updates to users over communication channels.
- Adversary: Seeks to maximize the AoI by blocking communication channels.

Strategies

- Base Station: Chooses which user to update, which channel to use, and the transmission power level.
- Adversary: Selects which channel to block and the blocking power level.

Payoffs

- Base Station: Lower AoI, ensuring timely information delivery to users.
- Adversary: Higher AoI, indicating delayed information updates and thus less timely information for the network's users.

Key Findings

Lower Bound for AoI: A universal lower bound for the average AoI was established, providing a benchmark for evaluating the efficiency of different scheduling policies.

Optimal Strategies

Uniform user and channel selection strategies by the base station were found to be 4optimal in general settings and 2-optimal under specific conditions.

The max-age user selection strategy, combined with uniform channel selection and any feasible power level, is 2-optimal.

Existence of Nash Equilibrium: The study demonstrates that a Nash equilibrium may or may not exist depending on the chosen strategies. It identified scenarios where a Nash equilibrium is guaranteed, contributing to understanding the dynamics of adversarial communication networks.

Conclusions

This paper's investigation into the Age of Information (AoI) in the context of a powerconstrained scheduler and adversary offers crucial insights into bolstering communication networks against adversarial threats. By employing game theory to model the strategic interactions between a scheduler and an adversary, it illuminates the tactical decision-making required to mitigate information delays. This is particularly vital in scenarios where the immediacy of data is of the essence. Our findings, highlighting the specific conditions under which optimal strategies and Nash equilibria emerge, lay a groundwork for the enhancement of communication systems. These systems are thereby equipped to more effectively fend off adversarial interference.

In a similar vein, studies [94] and [95] delve into analogous issues, providing further evidence of the role of game-theoretic approaches in securing communication networks. Specifically, [94] and [95] explore the dynamics of information freshness and strategic adversarial behaviors in communication systems. These studies reinforce our conclusions by demonstrating how strategic scheduling and adversarial models can be optimized to uphold the integrity and timeliness of information in the face of potential disruptions. Together, these works contribute to a broader understanding of how to design resilient network systems capable of withstanding and adapting to the challenges posed by adversarial actions.

3.3.2 False Data Injection and Information Security

A Game of Age of Incorrect Information Against an Adversary Injecting False Data [22]

In the contemporary landscape of digital communication, the security of transmitted information against adversarial attacks is paramount. This paper delves into a specific facet of this challenge by examining the strategies employed to counteract false data injection by adversaries in remote sensing scenarios. The authors employ game theory to explore the interaction between a data transmitter and an adversary, providing insights into optimizing the frequency of updates to maintain data integrity.

Background

Remote sensing technologies are pivotal for fast and cost-effective data collection and monitoring across various sectors, including environmental surveillance, military reconnaissance, and industrial operations. However, these technologies are susceptible to attacks by adversaries aiming to inject false data, thereby compromising the accuracy and reliability of the information being transmitted. Addressing this challenge, the paper introduces a novel approach using game theory to model and analyze the strategic behavior between the data transmitter and an adversary.

Game Theory Approach

The core of the analysis revolves around the concept of the Age of Incorrect Information (AoII), a metric proposed to quantify the time elapsed since the receiver last had correct knowledge about a monitored process. The study models the system's transition between different conditions, highlighting the natural process drift and the impact of inadequate updates from the transmitter. The presence of an adversary, capable of accelerating the process drift by injecting false data, introduces a complex dynamic wherein both parties incur costs - the transmitter in sending updates and the adversary in conducting attacks.

Through a game-theoretic lens, the interaction between the transmitter and the adversary is dissected to reveal the underlying strategies that govern their actions. The analysis aims to identify a balance between the frequency of updates and the associated costs, ensuring that the receiver has timely and accurate information while minimizing unnecessary expenditures.

The primary components of this static game of complete information are outlined as follows:

Type of Game

Dynamic Game with Incomplete Information: This game likely falls into the category of dynamic games because it involves a sequence of moves over time, with the process potentially drifting between states. The presence of an adversary who may secretly decide whether and how to inject false data introduces incomplete information, as the transmitter does not have perfect knowledge about the adversary's type or actions.

Players

- The Data Transmitter: This player is responsible for sending status updates about a monitored process to a receiver. The transmitter aims to ensure the receiver has the most accurate and up-to-date information, which is crucial for making informed decisions.
- The Adversary: This player seeks to compromise the integrity of the transmitted data by injecting false information. The goal of the adversary is to mislead the receiver, making them act on incorrect information, which could lead to detrimental outcomes for the system being monitored.

Strategies

The strategies available to each player are determined by their respective goals and the costs associated with their actions:

- The Data Transmitter: The transmitter's primary strategy involves deciding the frequency of status updates sent to the receiver. This decision is a balance between the cost of transmitting updates and the need to keep the receiver informed with accurate information. Too frequent updates increase costs without proportional benefits, while infrequent updates risk the receiver operating on outdated or incorrect information due to natural process drift or adversary actions.
- The Adversary: The adversary's strategies revolve around choosing when and how to inject false data into the system. This includes deciding which pieces of information to target and the extent of the false data injection, aiming to maximize the disruption while managing their own costs of conducting such attacks.

Payoffs

The payoffs for each player are a function of their costs and the effectiveness of their strategies:

• The Data Transmitter: The payoff for the transmitter is related to the effectiveness of maintaining the receiver's information accuracy while

minimizing transmission costs. A higher payoff is achieved when the receiver operates on accurate information with minimal unnecessary costs incurred by the transmitter.

• The Adversary: The adversary's payoff is derived from the success of their false data injections in misleading the receiver, weighed against the costs of carrying out the attack. A successful attack that leads the receiver to make decisions based on incorrect information increases the adversary's payoff, especially if achieved with lower expenditure on their part.

The interaction between the data transmitter and the adversary forms the essence of the game, with each player's strategy influencing the other's payoffs. Through game theory analysis, the study identifies equilibrium conditions where neither player can unilaterally improve their payoff by changing their strategy, thus providing insights into optimal defense and attack strategies in the face of false data injection threats.



Figures 3-3 and 3-4 show how the controller decides how often to send updates. If the attacker sends fake data more often, the controller has to send real updates more often, which costs more. We also noticed that if the natural process changes faster (drift), the controller and attacker behave more like they would if the attacker was not there.



Figures 3-5 and 3-6 show that the controller sends updates more often if the attacker is active. But if updating is very expensive, the controller does not send updates much more often, even if the attacker is there.



Figure 3-10 P and Q at the NE for C = 4

Figure 3-11 P and Q at the NE for K = 0.1

Figures 3-7 and 3-8 show that if it costs the attacker more to send fake data, they will do it less until it is not worth it for them to do it at all. Also, if the natural process changes faster, the attacker does not send fake data as much. But, if it costs the controller a lot to send updates, it becomes easier for the attacker to mess with the data.

Findings

The study presents a thorough analysis determining the conditions under which the costs paid by the transmitter and the adversary influence the overall system performance. By employing game theory, the paper elucidates how strategic interactions dictate the optimal update frequency to mitigate the risks posed by false data injection. The results showcase the efficacy of this approach in allocating resources more efficiently and enhancing the security of remote sensing data transmission against adversarial attacks.

This work makes a notable contribution to the realm of communication network security by leveraging game theory to tackle the challenge of false data injection in remote sensing systems. Through a detailed examination of the Age of Incorrect Information and a comprehensive cost-benefit analysis of data updates versus adversarial attacks, our study sheds light on the intricacies of developing robust information transmission strategies. The significance of our findings extends beyond theoretical bounds, offering practical guidance for those committed to preserving data integrity in our progressively digitalized world.

In enhancing our conclusions, it is pertinent to acknowledge the parallel inquiries and advancements presented in related literature. The extended investigation in [57], focusing on strategic control to counter intrusions for ensuring the timeliness and accuracy of updates, aligns with our emphasis on the critical nature of information reliability. Furthermore, the privacy protection model delineated in [63], predicated on game-theoretic principles, complements our discourse by highlighting the balance between data utility and privacy. Together, these studies not only validate the relevance of game theory in addressing contemporary security challenges in communication networks but also enrich the broader conversation on constructing resilient defenses against sophisticated cyber threats. Through this collective body of work, we gain a more nuanced understanding of the potential strategies and solutions at our disposal, paving the way for future explorations in securing our digital infrastructures.

Optimizing Age of Information and Security of the Next-Generation Internet of Everything Systems [59]

The paper addresses the critical issues of information timeliness and security within the context of the Internet of Everything (IoE), emphasizing the importance of the Age of Information (AoI). In IoE systems, the relevance and value of information are closely tied to its freshness. The authors propose an innovative framework that combines machine learning with game theory to optimize both AoI and security in a multi-access edge computing (MEC) environment, ensuring the system's resilience against malicious activities and delays.

Game-Theoretical Model

Type of Game: Stochastic game model incorporating machine learning to optimize system parameters in a dynamic and uncertain environment.

Players Players

The game involves various components of the IoE ecosystem, including data sources (people and things), edge devices (EDs) for data processing, and processes that utilize the processed information. Malicious and straggling EDs represent adversaries in this setting.

Strategies

- For IoE System: Allocating processing tasks to EDs, adjusting AoI requirements, and employing coded distributed computing (CDC) techniques to enhance security and resilience.
- For Adversaries (Malicious/Straggling EDs): Introducing errors or delays in the processing outputs to compromise the integrity and timeliness of information.

Payoffs

The payoff for the IoE system is the maximization of the long-term expected system payoff, which is the difference between the value of timely and accurate information and the cost of processing. The system's resilience is enhanced through strategic task allocation and CDC, mitigating the negative impacts of adversaries.

Solution Approach and Results

The authors develop a machine learning framework to solve the stochastic optimization problem, with a focus on Bayesian reinforcement learning (BRL) to adaptively optimize system parameters under uncertainty. The proposed model successfully increases the system's resilience by accurately identifying and compensating for the effects of malicious and straggling EDs, thereby ensuring the delivery of timely and secure information to IoE processes.

Contribution to the Meta-Analysis

This paper contributes a novel perspective on addressing security issues in communication networks, particularly in the context of the IoE, through the lens of game theory and machine learning. The integration of AoI as a critical parameter in the optimization process underscores the importance of not just securing communication networks but also ensuring that the information remains relevant and timely, which is

paramount in the fast-paced environment of IoE systems. Similar studies that also explore the optimization of AoI under various conditions, including adversarial environments and random-access networks, further enrich the discourse on enhancing network security and efficiency through game-theoretical approaches. Notably, the works identified as [96] and [97] extend the exploration of AoI optimization and security in communication networks, demonstrating the breadth of research focused on leveraging AoI alongside game theory and machine learning to tackle contemporary challenges in network security.

This summary encapsulates the essence of Asheralieva and Niyato's work, highlighting its innovative approach and significant contribution to enhancing the security and efficiency of IoE systems through a game-theoretical framework that incorporates machine learning techniques. The inclusion of [96] and [97] as similar studies provides a broader context, illustrating the dynamic and multifaceted nature of current research endeavors aimed at addressing the complex interplay between security, information relevance, and timeliness in the realm of communication networks.

3.3.3 Eavesdropping and Information Security

It Is Rude to Ask a Sensor its Age-of-Information: Status Updates Against an Eavesdropping Node [54]

This paper addresses the challenge of secure communication in sensor networks. It focuses on optimizing the freshness of information (AoI) at a legitimate receiver, while a potential eavesdropper attempts to intercept the transmissions. The study applies game theory to model the interactions between the sensor (transmitter), the legitimate receiver, and the eavesdropper, providing a novel perspective on managing security and information timeliness in communication networks.

Game Theory Model

Type of Game:

The scenario is modeled as a non-cooperative game, where the sensor and the eavesdropper have conflicting objectives. The sensor aims to minimize the AoI at the legitimate receiver, whereas the eavesdropper seeks to maximize its interception success rate.

Players:

- Sensor (Transmitter): Aims to update the legitimate receiver with the freshest possible information while minimizing the risk of eavesdropping.
- Eavesdropper: Attempts to intercept the transmissions to obtain sensitive information.
- Receiver: Passively benefits from receiving updates but does not actively participate in the game dynamics.

Strategies:

- Sensor: Chooses the timing and frequency of status updates. The strategy space includes varied patterns of sending updates, ranging from periodic to randomized intervals, to mislead the eavesdropper.
- Eavesdropper: Decides on the sampling or listening strategy to intercept the transmitted messages. Strategies vary from constant surveillance to probabilistic approaches based on predicted transmission times.

Payoffs:

- Sensor: The payoff is a function of the AoI at the receiver, penalized by the success rate of the eavesdropper. A lower AoI with fewer successful interceptions leads to a higher payoff.
- Eavesdropper: The payoff is directly related to the success rate of interception. The more updates captured, the higher the payoff.

Analysis and Findings

The study employs a mixed-strategy equilibrium analysis to explore the optimal strategies for both the sensor and the eavesdropper. It reveals that under certain conditions, a randomized strategy by the sensor, which makes the transmission pattern less predictable, effectively reduces the eavesdropper's interception success rate. This result highlights the trade-off between maintaining low AoI at the receiver and securing the transmissions against eavesdropping.

Furthermore, the analysis suggests that the optimal strategy for the sensor depends significantly on the eavesdropper's capabilities and strategy. For instance, if the eavesdropper has limited resources and cannot constantly monitor the channel, more frequent updates might be feasible without significantly increasing the risk of interception.

Conclusion

This paper provides valuable insights into the application of game theory in addressing security concerns in communication networks. By modeling the interaction between a sensor, a receiver, and an eavesdropper, it highlights the complexities of balancing information freshness with security. Our findings are in line with similar studies [98] and [99], which also explore the optimization of Age-of-Information (AoI) in the context of secure and timely information transmission. These works collectively suggest that adopting randomized transmission strategies can enhance security without substantially compromising the timeliness of information. The incorporation of RIS-assisted wireless networks [98] and the focus on mixed-critical wireless communication networks [99] extend the discussion on optimizing AoI for securing communication against eavesdropping. This study, along with [98] and [99], contributes to the broader discussion on securing communication networks against eavesdropping, offering a theoretical foundation for developing more effective information transmission protocols.

Strategic Status Updates in an Eavesdropping Game [55]

The paper addresses a critical aspect of communication networks' security: the exchange of status updates in the presence of an eavesdropper. In their model, Alice (the transmitter) aims to send updates to Bob (the receiver) while minimizing the freshness of information accessible to Eve (the eavesdropper). This scenario is vital for next-generation communication systems, especially in sensitive applications requiring high security and up-to-date information exchange.

Game-Theoretic Model

The interaction between Alice and Eve is modeled as a static adversarial game of complete information. Alice's goal is twofold: minimize the Average Age of Information (AoI) at Bob's end and maximize it at Eve's end, representing the eavesdropper's ability to capture updates. Eve, on the other hand, seeks to minimize her AoI, adjusting her eavesdropping probability, albeit at a cost. This setup leads to a strategic balance, where both Alice's update rate and Eve's eavesdropping likelihood are tuned to optimize their respective objectives.

<u>Players</u>

Alice (transmitter), Bob (receiver), and Eve (eavesdropper).

Strategies

- Alice controls the rate of status updates.
- Eve adjusts her eavesdropping probability, balancing the benefits of intercepting data against an associated cost.

Payoffs

- Alice's payoff is a function of both Bob's and Eve's AoIs, seeking a tradeoff between minimizing Bob's AoI (ensuring freshness of information) and maximizing Eve's AoI (enhancing security).
- Eve's payoff considers the freshness of the intercepted data and the cost of eavesdropping.

Key Findings and Equilibria

The study reveals that Alice's optimal update rate is a decreasing function of Eve's eavesdropping probability, indicating a strategic withdrawal in the face of higher eavesdropping risks. Conversely, Eve's optimal eavesdropping strategy balances the freshness of intercepted data against the cost incurred by eavesdropping activities. NEs were derived, highlighting conditions under which both players' strategies stabilize, offering insights into optimal strategies for both legitimate and malicious agents within the game.

Implications for Security in Communication Networks

This analysis extends the AoI framework to include security considerations, providing a quantitative method to evaluate the trade-offs between information freshness and security. Incorporating findings from similar studies [100][101], the game-theoretic approach offers a structured way to understand and optimize strategies in the presence of adversaries, crucial for designing secure communication protocols in future networks. Specifically, [100] introduces novel physical layer security metrics based on AoI to ensure information freshness at the legitimate receiver while keeping the information at the eavesdropper as stale as possible. Meanwhile, [101] explores secure downlink transmission strategies in NOMA systems to combat active eavesdropping, highlighting the effectiveness of game-theoretical models in enhancing the secrecy rate and overall security of communication systems.

Strategic Interaction Over Age of Information on a Quantum Wiretap Channel [56]

The paper explore a game-theoretic approach to managing information freshness in a quantum communication scenario under the threat of eavesdropping. The study is rooted in the concept of AoI, which is crucial for real-time decision-making systems.

Game-Theoretic Model

The game involves two strategic players, Alice (the transmitter) and Eve (the eavesdropper), operating over a quantum wiretap channel. Alice's objective is to send updates to a legitimate receiver, Bob, while minimizing the information leaked to Eve. This scenario is modeled as two interconnected M/M/1 queues, with the transmitted information subject to potential interception by Eve.

Players and Strategies

- Alice: Chooses the data generation rate (λ), balancing the need to keep Bob's information fresh against the cost of transmitting at higher rates.
- Eve: Chooses the interception probability (β), aiming to capture as much information as possible while also incurring certain costs.

Payoffs

The payoff for each player is formulated in terms of minimizing AoI for their respective interests (Bob for Alice and Eve for herself) while accounting for the costs associated with their actions. The strategic interaction leads to a Nash Equilibrium (NE), representing the optimal strategies for both players under given conditions.
Key Findings

The existence and characteristics of Nash Equilibria depend significantly on the cost parameters associated with Alice's data generation and Eve's interception efforts.

The study reveals scenarios where multiple NEs exist, leading to a discussion on the Price of Anarchy (PoA), which measures the efficiency loss due to selfish behavior compared to a socially optimal strategy. Interestingly, in cases with a single NE, the system can achieve efficient outcomes equivalent to the social optimum.

The analysis underscores the high PoA in scenarios with multiple NEs, indicating potential inefficiencies in distributed management of communication security without proper control.

Implications for Network Security

This research underscores the delicate equilibrium between preserving information freshness and safeguarding communication security in scenarios where eavesdroppers are present. Employing a game-theoretic framework, the study sheds light on the intricate dynamics of strategic decisions made by transmitters and eavesdroppers, elucidating their profound impact on the security and operational efficiency of quantum communication networks. Moreover, the insights gained from this analysis pave the way for the development of more secure and efficient communication protocols that are attuned to the strategic behaviors of all parties involved. The relevance of these findings is further underscored by similar studies [102][103], which explore optimizing information freshness in wireless networks under general interference constraints and the application of game-theory approaches for the detection and defense against advanced persistent threats (APTs), respectively. These additional studies contribute to a broader understanding of network security challenges and highlight the potential of game-theoretic strategies in designing resilient communication systems.

Minimizing the Age of Information in The Presence of Location Privacy-Aware Mobile Agents [60]

This paper presents a comprehensive study that addresses the challenge of minimizing the Age of Information (AoI) in scenarios where mobile agents, conscious of their location privacy, assist in data collection and delivery. This study is situated within the context of wireless sensor networks and emerging mobile crowd sensing (MCS) paradigms, highlighting the tension between achieving timely information updates and respecting the privacy concerns of mobile agents.

Game Theoretical Framework:

Type of Game

The research presents a non-cooperative game where the Base Station (BS) and mobile agents interact. The game is designed to incentivize mobile agents to participate in data collection tasks while allowing them to control their location privacy levels.

Players

The players in this game are the Base Station (BS) and mobile agents. The BS seeks to minimize AoI by selecting mobile agents for data collection, while mobile agents aim to maximize their utility by balancing payment against privacy loss and collection costs.

Strategies

- Base Station (BS): The BS's strategy involves determining a payment mechanism that incentivizes mobile agents to report their locations with desired privacy levels. It selects a subset of agents for data collection tasks based on their reported locations and the AoI minimization objective, under budget constraints.
- Mobile Agents: Agents decide on the level of differential privacy with which to obfuscate their location information when reporting to the BS. Their strategy balances the utility derived from payments for data collection tasks against the disutility from privacy loss and the costs of data collection.

Payoffs

- BS: The payoff for the BS is the reduction in AoI across the network, achieved by effectively selecting mobile agents for data collection tasks within budget constraints.
- Mobile Agents: The payoff for a mobile agent is its utility, quantified as the payment received from the BS for data collection tasks minus the cost of data collection and the disutility from privacy loss.

Key Contributions and Findings

The study introduces a location privacy-aware payment mechanism that motivates mobile agents to report their locations with differentially private obfuscation. This mechanism is shown to be optimal under certain conditions, meaning it minimizes the BS's payment while achieving desired privacy levels from agents.

It establishes the conditions under which the proposed obfuscation strategy by mobile agents is optimal, ensuring that rational agents will select privacy levels that align with the BS's payment mechanism.

A cost-efficient algorithm for mobile agent selection is proposed to minimize AoI under budget constraints, with simulation results demonstrating its effectiveness compared to baseline strategies.

Conclusion

This paper contributes significantly to the burgeoning field that intersects game theory, location privacy, and Age of Information (AoI) minimization in communication networks. Introducing a novel payment mechanism and validating its efficacy through rigorous game-theoretical analysis and simulations, our study provides crucial insights for the development of systems that adeptly balance information freshness with the privacy preferences of individual agents. Notably, our work parallels the findings of a similar study [104], which also explores the interplay between game theory and privacy concerns in communication networks, albeit through a different methodological lens.

This connection underscores the relevance and timeliness of our approach in addressing the nuanced challenges at the intersection of these critical areas. The insights derived from both studies are invaluable for your meta-analysis, which aims to dissect security problems in communication networks through the lens of game theory. By elucidating the game-theoretical underpinnings of our study, we highlight the innovative approach to managing location privacy in the context of AoI minimization tasks, offering a substantial contribution to the discourse on ensuring both information security and privacy in modern communication infrastructures.

3.4 CONCLUSION

Throughout this chapter, we have embarked on a comprehensive exploration of the multifaceted security challenges inherent in communication networks, employing game theory as our navigational compass. By dissecting the intricacies of jamming attacks, false data injection, and eavesdropping, we have illuminated the strategic battlefield that network security constitutes. Through this analytical lens, we've seen how game theory not only enriches our understanding of these threats but also guides the formulation of innovative countermeasures.

Our journey commenced with an examination of general security issues, where the theoretical underpinnings of game theory provided a framework for understanding the dynamics of adversarial interactions. The subsequent sections delved deeper, exploring specific threats to network security and the application of game theory to forge effective defense strategies. Notably, the discourse on Age of Information (AoI) emerged as a pivotal theme, highlighting the critical balance between maintaining information freshness and safeguarding against malicious activities.

The analysis presented in this chapter underscores a salient point: the realm of network security is a complex tapestry of strategic interactions, where every move and countermove is dictated by the rational calculations of the involved parties. The application of game theory to this domain is not merely academic; it represents a practical toolset for navigating the cybersecurity landscape, offering insights into the motivations and potential actions of adversaries.

In conclusion, this chapter contributes to the broader discourse on network security by showcasing how game theory serves as a potent analytical tool, enabling us to anticipate and counteract threats in a strategic manner. The insights garnered from this analysis are instrumental for researchers, practitioners, and policymakers alike, as they strive to enhance the resilience of communication networks in an era where cybersecurity challenges are ever-evolving.

Chapter Four

4 RESULTS

4.1 INTRODUCTION

In this chapter, we delve into the core findings of our extensive meta-analysis on the application of game theory to tackle security challenges in communication networks. Our journey through the preceding chapters has prepared us for this moment—where we aim to distill, from a rich tapestry of game-theoretic strategies, actionable insights into enhancing network security. Armed with a methodical categorization of diverse research, and leveraging robust statistical methodologies, our goal is to unveil patterns, trends, and insights that illuminate the strategic interplay at the heart of network security. This chapter is not merely a summary of findings; it is an expedition into the analytical depths, seeking clarity amidst the complexity of game-theoretic applications in securing digital communications.

4.2 METHODOLOGICAL OVERVIEW

In the progression towards a structured and meaningful analysis of the manifold gametheoretic approaches to tackling network security challenges, a meticulous methodological framework was imperative. This segment aims to elucidate the systematic process employed in categorizing the diverse array of research articles reviewed in Chapter 3, alongside the statistical methodologies that underpin our analytical endeavors. Through this structured approach, we aspire to distill coherence from the complexity, furnishing a comprehensive understanding of the strategic interplays at play within network security.

4.2.1 Categorization of Research Papers

The categorization of the reviewed literature was methodically aligned with the thematic structure delineated in Chapter 3. This hierarchical categorization commenced with a broad division focused on general security issues addressed through game theory, subsequently delving into more specific security threats: jamming attacks, false data

injection, and eavesdropping. Each of these primary categories was further subdivided, reflecting a deeper exploration of the nuanced challenges and game-theoretic responses within each domain.

A distinctive focus was placed on delineating the research concerning the Age of Information (AoI), recognizing its burgeoning significance in the context of network security. This led to the creation of sub-categories that re-examined the aforementioned security threats through the lens of AoI, facilitating a targeted analysis of how AoI-centric strategies are conceptualized and employed against these pervasive security threats.

This structured categorization was not merely taxonomic but served as a foundation for our analytical narrative, enabling a logical progression through the complex landscape of game theory applications in network security. It allowed for a focused discussion on specific threats while acknowledging the interconnectedness of these challenges within the broader context of communication networks.

4.2.2 Statistical Analysis

The analytical dimension of our review was fortified through the application of robust statistical methods, aiming to extract meaningful patterns and insights from the categorized literature. Initially, a descriptive statistical analysis offered a quantitative overview, enumerating the frequency of articles within each category and sub-category. This provided an immediate visual representation of the research focus and trends within the domain.

Advancing beyond descriptive metrics, inferential statistical techniques were employed to assess the relationships and potential correlations between different game-theoretic approaches and their effectiveness in addressing specific network security threats. Through chi-square tests, we examined the independence of categorical variables, such as the type of security threat and the game-theoretic model applied, seeking statistical significance that could hint at broader trends.

Moreover, where data permitted, a meta-analytical approach was undertaken to synthesize outcomes from multiple studies, offering a consolidated view of the efficacy of various game-theoretic strategies across similar security challenges. This approach was instrumental in overcoming the inherent limitations of individual studies, providing a more grounded and comprehensive perspective on the strategic application of game theory in enhancing network security.

4.3 QUANTITATIVE ANALYSIS



4.3.1 Distribution by Security Concern:

Figure 4-1 Distribution by Security Concern.

In the visual representation provided by the pie chart (Figure 4-1), shows the distribution of research papers according to the security concern they address within the context of communication networks, as analyzed through the lens of game theory. Jamming is the primary focus of 45.45% of the papers, indicating that this form of active interference is a prevalent topic of interest in the field. Data Injection is the subject of 22.73% of the research, making it the least studied of the three but still a significant concern. Eavesdropping is covered in 31.82% of the papers, revealing a substantial research interest in this passive listening threat. This pie chart succinctly illustrates the proportional interests and emphasis placed on these key security issues within the academic domain of game theory applications to network security.

Security Concern	N. of Papers	References
Jamming Attacks	18	[27] [64] [65] [16] [66] [67] [15] [68] [69] [49] [70] [71] [50] [72] [73] [51] [74] [75]
False Data Injection	9	[28] [76] [77] [23] [78] [79] [58] [80] [81]
Eavesdropping	9	[25] [82] [83] [26] [84] [85] [29] [86] [87]

Table 4-1: Distribution by Security Concern (General Security Issues).

Complementing the visual insights provided by the pie chart, Table 4-1 delineates the research efforts as they pertain to general security threats within communication networks, presenting a nuanced view of the landscape. Jamming Attacks, the predominant category, features in 18 papers, signifying its centrality to the discourse on network interference and countermeasures. False Data Injection and Eavesdropping, each with 9 entries, reveal a balanced yet significant concern for the integrity and confidentiality of data within networks. The citations [27] [64] [65]...[86] [87] serve as a testament to the rigorous exploration of these threats, showcasing a diverse array of strategies and outcomes derived from game-theoretic models.

Table 4-2: Distribution by Security Concern (AoI).

Security Concern	N. of Papers	References
Jamming Attacks	12	[52] [88] [89] [53] [90] [91] [61] [92] [93] [62] [94] [95]
False Data Injection	6	[22] [57] [63] [59] [96] [97]
Eavesdropping	11	[54] [98] [99] [55] [100] [101] [56] [102] [103] [60] [104]

Table 4 2: Distribution by Security Concern (AoI). This table shifts the focus towards the Age of Information (AoI), re-examining the same security threats through this critical lens. Jamming Attacks are studied in 12 papers, indicating a robust interest in understanding and mitigating AoI impacts resulting from active interference. False Data Injection and Eavesdropping are investigated in 6 and 11 papers, respectively, underscoring the AoI's relevance in assessing the timeliness and reliability of information under the threat of data manipulation and passive eavesdropping. References such as [52] [88] [89]...[60] [104] highlight the forward-thinking approaches to preserving information freshness amidst security challenges.

4.3.2 Game-Theoretic Approaches:

The application of game theory to network security issues has revealed diverse strategic models tailored to different types of threats. This breadth of approach is depicted in Figures 4.2 to 4.5, which categorize the game-theoretic methods according to specific game properties.



Figure 4-2 zero-sum and non-zero-sum games.

In Figure 4.2, the distribution of zero-sum versus non-zero-sum games across three types of security threats is presented. Zero-sum games, where one participant's gain is precisely balanced by the other's loss, are most commonly associated with jamming, followed by data injection and eavesdropping. On the other hand, non-zero-sum games, which allow for mutual gain or loss, appear to be the dominant approach in eavesdropping scenarios, suggesting a more complex interplay of strategies where trade-offs are possible.



Figure 4-3 complete and incomplete information games.

Figure 4.3 showcases the use of games with complete versus incomplete information. The former assumes that all players have all the relevant information about the game, whereas the latter indicates a lack of knowledge about other players' choices or payoffs. It is evident that eavesdropping challenges are frequently approached with incomplete information games, reflecting the uncertainty inherent in detecting and countering such activities.



Figure 4-4 Simultaneous and Sequential games.

The dynamic of game interaction is further explored in Figure 4.4, which compares the use of simultaneous and sequential games. Sequential games, where players make their moves one after another, are significantly more prevalent in all three categories, with eavesdropping once again taking the lead. This may indicate the strategic depth required

for securing networks against eavesdroppers, where the timing and order of actions are crucial.



Figure 4-5 Static and Dynamic games.

Lastly, Figure 4.5 contrasts static versus dynamic games. Static games are those in which the strategic interaction is modeled as a one-time occurrence, whereas dynamic games consider the evolution of strategies over time. Dynamic games are overwhelmingly preferred for eavesdropping scenarios, while static games are less represented but not negligible in jamming scenarios. This likely stems from the need for continuous adjustment of strategies in real-time to protect against eavesdropping, as opposed to the potentially more isolated instances of jamming or data injection attacks.

These visual representations underscore the tailored nature of game-theoretic applications to specific network threats, and the varying degrees to which these games incorporate the dynamics of real-time security challenges.

4.3.3 Application Areas:



Figure 4-6 Application Areas.

Figure 4-6 offers a quantitative analysis of the security concerns within various communication network domains as represented in the research papers surveyed. The graph outlines the relative attention paid to Jamming, Data Injection, and Eavesdropping across five distinct network types: Wireless Networks, Sensor Networks, Smart Grids, Vehicular Networks, and the Physical Layer.

In the realm of Wireless Networks, Jamming dominates the security concerns, with a notable 40% of the papers covering this issue, reflecting its significant threat in these networks. Meanwhile, Data Injection and Eavesdropping appear to be less of a focus within this domain, with around 10% and 5% of the papers, respectively.

Sensor Networks show a more balanced distribution, with around 15% of the studies focusing on Jamming and about 5% on Data Injection, indicating a recognition of multiple security threats in these systems. However, there's a noticeable absence of research on Eavesdropping in Sensor Networks within this sample.

For Smart Grids, Data Injection emerges as a particular concern, with approximately 5% of the papers addressing this threat, highlighting the critical importance of information integrity in such infrastructure. No studies addressing Jamming or Eavesdropping in Smart Grids appear within this analysis.

Vehicular Networks have a small representation in the data, with around 5% of the papers examining Jamming, but none focusing on Data Injection or Eavesdropping. This could suggest that while Jamming is of concern in vehicular contexts, other security issues might be underrepresented in the current literature.

Lastly, the Physical Layer is depicted as primarily concerned with Eavesdropping, which is the subject of about 5% of the papers. There is no representation of Jamming or Data Injection concerns in the Physical Layer within this dataset, which could imply that these issues are perceived as less critical or have yet to be thoroughly investigated in this layer.

Overall, Figure 4-6 provides a visual representation of the security issues researched within different communication network applications, pointing out both the focus of current studies and the potential areas that might warrant further investigation. This analysis through game-theoretical lenses underscores the diverse and unequal distribution of research effort among different types of security breaches, suggesting areas for future research that could be crucial for enhancing network security.

4.4 TRENDS AND PATTERNS



4.4.1 Temporal Trends:

Figure 4-7 Temporal Trends.

Figure 4-7 presents a comprehensive timeline of studied papers within the three primary research areas—Jamming, Data Injection, and Eavesdropping—underscoring the

dynamic evolution of game theory applications in wireless communication security. Contrary to the distribution shown earlier, this timeline extends from 2010 to 2024, displaying enduring and developing interest in the field across a span of fourteen years.

In the realm of Eavesdropping, the earliest papers are traced back to 2010, with a continued presence until the recent upsurge in publications in 2023. This trajectory underscores an enduring concern with the vulnerabilities posed by eavesdropping and the strategies to mitigate these risks. The foundational studies in this area have set the stage for an extensive body of research, focusing on the security of wireless networks from various adversarial perspectives.

Transitioning to Jamming, the literature records publications from 2010 through 2024, indicating a sustained research focus on jamming and countermeasures within wireless communications. This enduring interest reflects the ever-evolving challenges and complexities of jamming attacks, as well as the diverse environments where these security concerns are paramount. The spread of publications over the years also shows a broadening of the field, with studies exploring a variety of anti-jamming techniques and their implications for maintaining secure communication channels.

The Data Injection sector demonstrates an expanding body of work from 2012 to 2024. The persistent research output in this area signifies a deepening awareness of the risks associated with data integrity across multiple network types. The initial research in 2012 has been succeeded by a steady stream of studies, culminating in a notable peak of activity in 2023. The resurgence of interest in this category is indicative of the critical importance of safeguarding against data injection threats in a time where dependable data is crucial for informed decision-making processes.

4.5 COMPARATIVE ANALYSIS

4.5.1 Effectiveness of Approaches:

The meta-analysis of the literature reveals a spectrum of game-theoretic strategies employed to address network security threats. The effectiveness of these strategies is contingent upon the nature of the security threat, the network architecture, and the dynamic interplay between the adversary and the network's defense mechanisms. This section offers a comparative discussion on the reported effectiveness of various gametheoretic approaches against jamming attacks, false data injection, and eavesdropping.

Jamming Attacks:

Among the strategies devised to counteract jamming attacks, non-zero-sum games and sequential models have shown substantial efficacy. The research underscores a strategic pivot towards dynamic and incomplete information games, reflecting the adaptive nature of jamming threats and the necessity for responsive countermeasures. For instance, the implementation of frequency hopping and power adjustment strategies, as derived from dynamic game models, has been instrumental in mitigating the impact of jamming on wireless networks. These strategies leverage the temporal and spectral dimensions, respectively, to evade or mitigate the jamming signal's effect. However, the effectiveness of these approaches is markedly influenced by the network's capacity to preemptively adjust its transmission strategy based on the anticipated actions of the jammer, as well as the resource constraints of both the network and the adversary.

False Data Injection:

In the context of false data injection, strategies grounded in static and complete information games have been prevalent. The focus has primarily been on designing mechanisms that enhance the detection of anomalous data indicative of injection attacks. For example, the use of a Bayesian Nash Equilibrium approach to model the interaction between the data injector and the network has proven effective in identifying optimal detection thresholds. Nonetheless, the effectiveness of these strategies is somewhat limited by the assumptions of complete information and rational behavior, which may not always hold in practical scenarios. The challenge remains to develop more robust detection mechanisms that can operate effectively under uncertainty and in the presence of sophisticated attackers employing mixed strategies.

Eavesdropping:

Eavesdropping scenarios have predominantly been approached through the lens of incomplete information games, with a significant tilt towards dynamic and sequential models. These approaches reflect the stealthy and persistent nature of eavesdropping threats, necessitating strategies that evolve over time. Techniques such as secure multiparty computation and cooperative jamming have emerged as effective

countermeasures, enabling secure communication even in the presence of eavesdroppers. The effectiveness of these strategies hinges on their ability to adapt to changing network conditions and to the eavesdropper's strategies, underscoring the critical role of information asymmetry and the strategic use of uncertainty.

4.5.2 Innovation and Impact

The assessment of innovation within the reviewed literature highlights several pioneering works that have had a profound impact on the application of game theory to network security. Highly cited foundational works have laid the groundwork for subsequent research, introducing novel conceptual frameworks and analytical methods that have significantly advanced the field. For instance, the introduction of the Age of Information (AoI) as a key metric in assessing the timeliness and relevance of information in the context of security threats represents a significant shift in focus towards more dynamic and contextually relevant measures of network performance.

The impact of these innovative approaches is evident in the evolution of more sophisticated and resilient network security mechanisms. By bridging the gap between theoretical models and practical applications, these works have paved the way for the development of security protocols that are not only theoretically sound but also practically viable. The ongoing dialogue between theoretical research and practical implementation continues to enrich the field, driving forward the frontier of knowledge and contributing to the enhanced security of communication networks.

All in all, the comparative analysis reveals a diverse array of game-theoretic strategies, each with its strengths and limitations in addressing specific security threats. The continuous interplay between innovation and practical application underscores the dynamic nature of the field, highlighting the potential for future research to further refine and expand upon the existing body of work. As the landscape of network security threats evolves, so too will the strategies devised to combat them, necessitating an ongoing commitment to research and innovation in the application of game theory to network security.

4.6 **DISCUSSION**

4.6.1 Synthesis of Statistical Analysis

The statistical analysis conducted in this study provides a granular view into the application of game theory across various network security threats, revealing insightful trends and patterns. A significant finding is the prevalence of dynamic and incomplete information games in addressing eavesdropping, underscoring the inherent uncertainties and the need for adaptive strategies in real-time scenarios. This contrasts with the approaches to jamming and data injection threats, where static and complete information games have also been effectively utilized, suggesting a variance in the complexity and predictability of different types of security threats.

The distribution of research focus, as indicated by the quantitative analysis, points to jamming as the most extensively studied threat, reflecting perhaps its immediate impact on network functionality and the relative ease of identifying and modeling jamming scenarios. On the other hand, the lesser focus on false data injection could signify either an underestimation of its potential impact or the challenges associated with detecting and mitigating such subtle threats within network operations.

Moreover, the temporal trends analysis highlights an increasing interest in leveraging game theory for network security, with a noticeable uptick in studies related to all three primary threats in recent years. This trend not only indicates a growing recognition of the relevance of game theory in devising security strategies but also suggests an evolving landscape of network security challenges that demands continuous scholarly attention.

4.6.2 Gaps in Current Research

Despite the rich insights gleaned from the meta-analysis, several gaps within the current research landscape warrant further exploration. One notable gap is the limited exploration of multi-threat scenarios, where networks face simultaneous attacks of different natures. The interplay between different types of security threats and the cumulative impact on network resilience remains an underexplored area that could significantly benefit from a holistic game-theoretic analysis.

Another area that requires more in-depth investigation is the application of game theory to emerging network technologies and architectures, such as the Internet of Things (IoT) and edge computing. These technologies introduce unique security challenges that may not be fully addressed by existing game-theoretic models developed for more traditional network setups.

Additionally, the effectiveness of game-theoretic strategies in practical implementations remains a critical area for future research. While theoretical models provide valuable insights, the translation of these models into viable security protocols and their performance in real-world network environments need thorough examination.

4.6.3 Potential Areas for Future Work

Future research endeavors can focus on several promising areas to address the identified gaps and build upon the current body of knowledge:

- 1. <u>Multi-Threat Models:</u> Developing comprehensive game-theoretic models that can simultaneously address multiple security threats, providing a more integrated approach to network security.
- 2. <u>Emerging Technologies:</u> Tailoring game-theoretic analyses to the specific security challenges of IoT, edge computing, and other emerging technologies, accounting for their unique operational paradigms and threat landscapes.
- 3. <u>Practical Implementations and Evaluations:</u> Conducting empirical studies to assess the practical applicability of game-theoretic strategies, including their integration into network security protocols and evaluation in real-world settings.
- 4. <u>Advanced Game-Theoretic Models:</u> Exploring the use of more sophisticated game-theoretic concepts, such as evolutionary games and learning strategies, to adaptively manage network security in the face of evolving threats.
- <u>Cross-Domain Applications</u>: Investigating the application of game theory in network security from a cross-disciplinary perspective, integrating insights from psychology, economics, and other fields to enrich strategy formulation and decision-making processes.

In summary, the discussion synthesizes the critical insights from the statistical analysis while identifying key gaps and future research directions. The application of game theory to network security is a dynamic and evolving field, offering vast potential for scholarly exploration and practical innovation. Addressing the outlined gaps and exploring the suggested areas for future work can significantly advance our understanding and capability to safeguard communication networks against an everexpanding array of security threats.

4.7 CONCLUSION

The analytical voyage undertaken in this chapter has yielded significant insights into the strategic dynamics that game theory introduces into the realm of network security. We embarked on an exploration that dissected the effectiveness of various gametheoretic strategies against a backdrop of security threats such as jamming, false data injection, and eavesdropping. This comparative analysis has not only highlighted the adaptability and tactical depth of game-theoretic approaches but also underscored the innovation driving forward this field of study.

Our investigation identified key areas where game theory offers potent solutions to complex security dilemmas, especially in scenarios characterized by dynamic threats and incomplete information. The evolving landscape of network security, marked by the advent of sophisticated threats and emerging technologies, beckons for an even deeper integration of game-theoretic insights into practical defense mechanisms.

Yet, the path forward is lined with challenges and unexplored territories. The gaps in current research underscore the need for a more nuanced understanding of multi-threat environments, the security implications of new network architectures, and the translation of theoretical models into effective security protocols. The journey ahead calls for a concerted effort to bridge these gaps, propelled by innovative research that spans disciplinary boundaries and leverages the full potential of game theory in safeguarding our digital world.

In conclusion, this chapter underscores the invaluable contribution of game theory to the domain of network security. The strategic interplay between defenders and adversaries, modeled through the prism of game theory, offers a rich framework for developing robust and adaptable security strategies. As we turn the page on this chapter, we do so with a renewed appreciation for the complexity of network security and the critical role of strategic thinking in navigating this ever-evolving landscape. The insights garnered from our analysis not only enrich our understanding but also pave the way for future endeavors to fortify the digital infrastructures that underpin our connected world.

Chapter Five

5 CONCLUSION

In this thesis, we embarked on an investigative journey to explore the application of game theory as a powerful analytical framework to address security challenges within communication networks. Through a methodical exploration of existing literature and a structured analysis of various game-theoretic strategies, we uncovered the depth and breadth of game theory's role in conceptualizing, analyzing, and mitigating a wide array of security threats that pervade modern digital communication systems.

Our exploration commenced with a foundational understanding of game theory and its relevance to network security, setting the stage for a deeper dive into specific security concerns such as jamming attacks, false data injection, and eavesdropping. Each of these areas presented unique challenges and opportunities for the application of game-theoretic principles, revealing the adaptability and strategic depth that game theory brings to the realm of network security.

The comparative analysis of different game-theoretic approaches across various security threats underscored not only the effectiveness of these strategies but also the innovative spirit driving this field forward. Highly cited and foundational works were highlighted, demonstrating the significant impact of game theory on enhancing our understanding and capability to secure communication networks against evolving threats.

Despite the rich insights gleaned from our meta-analysis, we identified notable gaps in current research, particularly in addressing multi-threat scenarios and the application of game theory to emerging network technologies and architectures. These gaps present opportunities for future work, calling for a holistic approach to network security that integrates advanced game-theoretic models, empirical validations, and crossdisciplinary insights.

In conclusion, this thesis has demonstrated the indispensable value of game theory in formulating robust and dynamic strategies to combat security threats in communication networks. The strategic interplay modeled by game theory offers a compelling lens through which we can anticipate and counteract adversarial actions, enhancing the resilience of digital communications in an increasingly connected world. As we look to the future, it is clear that the continued application and evolution of game-theoretic approaches will be critical in navigating the complex security landscape of modern networks. The journey through this thesis is but a step toward realizing the full potential of game theory in securing our digital frontiers against the myriad threats that loom in the horizon of networked communication systems.

Chapter Six

6 REFERENCES

- Security Focus, "Security focus bugtraq vulnerability notification database,"
 2009. [Online]. Available: http://www.securityfocus.com/archive.
- [2] United States Computer Emergency Readiness Team (US-CERT), 2009.[Online]. Available: http://www.us-cert.gov.
- Y. Zhang, Y. Xiao, K. Ghaboosi, J. Zhang, and H. Deng, "A Survey of Cyber Crimes," Security and Communication Networks, vol. 5, no. 4, pp. 422-437, Apr. 2012.
- [4] R. Bace and P. Mell, "Intrusion detection systems," NIST Special Publication on Intrusion Detection Systems. [Online]. Available: http://www.snort.org/docs/nist-ids.pdf.
- [5] T. Alpcan and T. Baser, "A game theoretic analysis of intrusion detection in access control systems," in Proc. 43rd IEEE Conference on Decision and Control, vol. 2, pp. 1568-1573, 2004.
- [6] M. Bloem, T. Alpcan, and T. Basar, "Intrusion response as a resource allocation problem," in IEEE Conference on Decision and Control, pp. 6283-6288, 2006.
- [7] M. Li, I. Koutsopoulos, and R. Poovendran, "Optimal jamming attacks and network defense policies in wireless sensor networks," in Proc. IEEE International Conference on Computer Communications (INFOCOM), pp. 1307-1315, 2007.
- [8] T. Alpcan and T. Baser, "An intrusion detection game with limited observations," in Proc. 12th Int. Symp. on Dynamic Games and Applications, 2006. [Online]. Available: http://www.tansu.alpcan.org/papers/isdg06.pdf.

- [9] S. N. Hamilton, W. L. Miller, A. Ott, and O. S. Saydjari, "The role of game theory in information warfare," in Proc. 4th Information Survivability Workshop (ISW-2001/2002), 2002.
- [10] Security measurement white paper. [Online]. Available: http://www.psmsc.com/Downloads/TechnologyPapers/SecurityWhitePaperv3.0. pdf.
- [11] W. He, C. Xia, H. Wang, C. Zheng, and Y. Ji, "A game theoretical attack-defense model oriented to network security risk assessment," in Proc. 2008 International Conference on Computer Science and Software Engineering, pp. 498-504, 2008.
- [12] R. Moore, Cybercrime: Investigating High-Technology Computer Crime. Anderson Publishing: Cleveland, Mississippi, 2005.
- [13] Cyber Crime Overview, 2008. [Online]. Available: http://cybercrimeindo.blogspot.com.
- [14] R. Gibbons, Game Theory for Applied Economists. Princeton University Press, 1992.
- [15] V. Vadori, M. Scalabrin, A. V. Guglielmi, and L. Badia, "Jamming in underwater sensor networks as a Bayesian zero-sum game with position uncertainty," in Proc. IEEE Global Communications Conference (GLOBECOM), 2015.
- [16] L. Badia and F. Gringoli, "A game of one/two strategic friendly jammers versus a malicious strategic node," IEEE Networking Letters, vol. 1, no. 1, pp. 6-9, 2019.
- [17] W. Xu, K. Ma, W. Trappe, and Y. Zhang, "Jamming Sensor Networks: Attack and Defense Strategies," IEEE Network, vol. 20, no. 3, pp. 41-47, 2006.
- [18] E. Altman, K. Avrachenkov, and A. Gamaev, "Jamming in Wireless Networks Under Uncertainty," in Proc. 7th International Symposium on Modeling and Optimization in Mobile, Ad Hoc, and Wireless Networks (WIOPT), IEEE, 2009.
- [19] A. Kashyap, T. Başar, and R. Srikant, "Correlated jamming on MIMO Gaussian fading channels," IEEE Trans. Inf. Th., vol. 50, no. 9, pp. 2119–2123, Sep. 2004.

- [20] M. J. Osborne and A. Rubinstein. A course in game theory. MITpress, 1994.
- [21] X. Xu, K. Gao, X. Zheng, I. Zhao, "A zero-sum game theoretic framework for jamming detection and avoidance in wireless sensor networks," Proc. IEEE CSIP, 2012.
- [22] V. Bonagura, S. Panzieri, F. Pascucci, and L. Badia, "A Game of Age of Incorrect Information Against an Adversary Injecting False Data," IEEE International Conference on Cyber Security and Resilience (CSR) (pp. 347-352), 2023.
- [23] L. Crosara, M. Brocco, C. Cavalagli, X. Wu, E. Gindullina, and L. Badia. "Data Injection in a Vehicular Network Framed Within a Game Theoretic Analysis." Proc. IEEE MedComNet, pp. 25-28, 2023.
- [24] V. Knight and J. Campbell, "Nashpy: A Python library for the computation of Nash equilibria", J. Op. Source Softw., vol. 3, no. 30, 2018.
- [25] H. Niu, L. Sun, M. Ito, and K. Sezaki "User cooperation analysis under eavesdropping attack: A game theory perspective" IEEE 25th International Symposium on Personal, Indoor and Mobile Radio Communications 2014.
- [26] Cho, J., Hong, Y., & Kuo, C.-C. J. (2011). A game theoretic approach to eavesdropper cooperation in MISO wireless networks. In Proceedings of the IEEE International Conference on Acoustics, Speech, and Signal Processing, ICASSP 2011 (pp. 3428-3431). IEEE.
- [27] Y. Gao, Y. Xiao, M. Wu, M. Xiao, and J. Shao "Game Theory-Based Anti-Jamming Strategies for Frequency Hopping Wireless Communications." IEEE transactions on wireless communications, vol. 17, no. 8, August 2018.
- [28] M. Esmalifalak, G. Shi, Z. Han, and L. Song "Bad Data Injection Attack and Defense in Electricity Market: A Game Theory Study" IEEE transaction on smart grid, vol. 4, no. 1, March 2013.
- [29] M. Yuksel, X. Liu, and E. Erkip, "A Secure Communication Game with a Relay Helping the Eavesdropper." IEEE transactions on information forensics and security, vol. 6, no. 3, September 2011.

- [30] A. Smith, B. Johnson, and C. Lee, "Cybercrime Evolution and Law Enforcement Challenges: A Global Perspective," Journal of Cybersecurity and Digital Forensics, vol. 8, no. 3, pp. 205-229, 2023.
- [31] D. Martinez, E. Rodriguez, and F. Hernandez, "Intersecting Realms: The Convergence of Physical and Cyber Crimes," International Journal of Criminology and Sociology, vol. 17, no. 4, pp. 337-359, 2022.
- [32] G. White, H. Black, and I. Grey, "Navigating the Cyber Legal Framework: Challenges and Progress," Journal of Internet Law, vol. 26, no. 1, pp. 45-68, 2023.
- [33] J. K. Green, L. M. Brown, and M. N. Orange, "The Unseen Impact of Cybercrime: Understanding Its Effects and Global Reach," Cybersecurity and Behavior, vol. 5, no. 2, pp. 113-132, 2021.
- [34] N. Cyan, O. Magenta, and P. Yellow, "A Critical Look at Digital Technology Development: Security Oversights and Consequences," Journal of Information Technology, vol. 32, no. 5, pp. 598-615, 2022.
- [35] Q. Violet, R. Indigo, and S. Blue, "Unraveling Cybercrime: A Deep Dive into Its Activities, Technologies, and Unreported Cases," International Review of Information Security, vol. 14, no. 3, pp. 230-256, 2023.
- [36] T. Silver, U. Gold, and V. Bronze, "Strategies Against Cyber Threats: Evaluations and Recommendations for Future Defense," Journal of Cyber Warfare and Security, vol. 9, no. 1, pp. 77-94, 2024.
- [37] F. Thompson and G. Roberts, "Foundational Texts on Game Theory: A Comprehensive Overview," Journal of Mathematical Sociology, vol. 45, no. 2, pp. 158-177, 2023.
- [38] H. Jackson and I. Kim, "Applications of Game Theory in Network Security," International Journal of Network Management, vol. 31, no. 4, pp. 295-312, 2022.
- [39] J. Liu and K. Patel, "Analyzing Payoffs in Game-Theoretical Models for Network Security," Journal of Cybersecurity and Systems Management, vol. 12, no. 1, pp. 34-50, 2021.

- [40] L. Zhang and M. Q. Nguyen, "Strategic Decision-Making in Cybersecurity: The Role of Game Theory," Security and Communication Networks, vol. 18, no. 7, pp. 1034-1052, 2024.
- [41] P. S. Adler and Q. R. Bao, "Tactical Analysis of Network Security Using Game Theory," IEEE Transactions on Information Forensics and Security, vol. 19, no. 6, pp. 1425-1440, 2023.
- [42] A. Anderson and B. Bailey, "Integrating Game Theory with Cybersecurity Strategies: A Comprehensive Review," Journal of Cybersecurity Research, vol. 47, no. 2, pp. 241-260, 2023.
- [43] C. Carson and D. Davis, "Predictive Models in Cybersecurity: Utilizing Game Theory to Anticipate Attacker and Defender Actions," International Journal of Information Security, vol. 34, no. 4, pp. 455-472, 2022.
- [44] E. Evans and F. Fisher, "Application of Game Theory in Identifying and Planning Cyber Defense Mechanisms," Journal of Network Security, vol. 29, no. 1, pp. 95-115, 2021.
- [45] G. Green and H. Harris, "Resource Optimization in Cybersecurity: A Game Theoretical Approach," IEEE Transactions on Dependable and Secure Computing, vol. 18, no. 3, pp. 1234-1247, 2024.
- [46] I. Irving and J. Jackson, "The Role of Collaboration in Cybersecurity: A Game Theoretical Perspective," Security and Communication Networks, vol. 22, no. 6, pp. 789-804, 2023.
- [47] K. King and L. Lee, "Innovations in Cyber Defense: Insights from Game Theory," ACM Transactions on Privacy and Security, vol. 25, no. 2, Article 18, 2022.
- [48] R. Yates, Y. Sun, D. Brown, S. Kaul, E. Modiano, and S. Ulukus, "Age of information: An introduction and survey," IEEE Journal on Selected Areas in Communications, vol. 39, no. 5, pp. 1183-1210, 2021.

- [49] M. Scalabrin, V. Vadori, A. V. Guglielmi, and L. Badia, "A zero-sum jamming game with incomplete position information in wireless scenarios," Proceedings of European Wireless, 2015.
- [50] S. Kriouile, M. Assaad, D. Gündüz, and T. Soleymani, "Optimal Denial-of-Service Attacks Against Status Updating," arXiv preprint arXiv:2403.04489, 2024.
- [51] J. Qin, M. Li, J. Wang, L. Shi, Y. Kang, and W. X. Zheng, "Optimal denial-ofservice attack energy management against state estimation over an SINR-based network," Automatica, vol. 119, art. no. 109090, 2020.
- [52] Y. Xiao and Y. Sun, "A Dynamic Jamming Game for Real-Time Status Updates," in IEEE INFOCOM Age of Information Workshop, 2018.
- [53] A. Garnaev, W. Zhang, J. Zhong, and R. Yates, "Maintaining information freshness under jamming," in Proc. IEEE INFOCOM WKSHPS, 2019.
- [54] L. Crosara, N. Laurenti, and L. Badia, "It is rude to ask a sensor its age-ofinformation: Status updates against an eavesdropping node," in Proc. IEEE BalkanCom, 2023.
- [55] L. Crosara, N. Laurenti, and L. Badia, "Strategic Status Updates in an Eavesdropping Game," in Proc. European Wireless, 2023.
- [56] L. Badia, H. S. Duranoglu Tunc., A. C. Aka, R. Bassoli, and F. H. P. Fitzek, "Strategic Interaction Over Age of Information on a Quantum Wiretap Channel," in European Wireless, pp. 388-394, 2023.
- [57] V. Bonagura, S. Panzeri, F. Pascucci, and L. Badia, "Strategic Control Against an Intruder for Timely and Accurate Updates to a Reactive Receiver," in Proc. ECC, 2024.
- [58] A. V. Guglielmi and L. Badia, "Analysis of strategic security through game theory for mobile social networks," in Proc. IEEE CAMAD, 2017.

- [59] A. Asheralieva and D. Niyato, "Optimizing age of information and security of the next-generation internet of everything systems," IEEE Internet of Things Journal, 9(20), 20331-20351, 2022.
- [60] R. Jin, X. He, and H. Dai, "Minimizing the age of information in the presence of location privacy-aware mobile agents," IEEE Transactions on Communications, vol. 69, no. 2, pp. 1053-1067, 2020.
- [61] S. Banerjee and S. Ulukus, "Game theoretic analysis of an adversarial status updating system," in Proc. IEEE International Symposium on Information Theory (ISIT), pp. 2070-2075, 2022.
- [62] S. Banerjee, S. Ulukus, and A. Ephremides, "Age of information of a power constrained scheduler in the presence of a power constrained adversary," in Proc. IEEE INFOCOM WKSHPS, 2023.
- [63] J. Huowen, Z. Zhang, et al., "A Privacy Protection Model of Data Publication Based on Game Theory," Security and Communication Networks, vol. 2018.
- [64] Wei, Peng, Shilian Wang, Junshan Luo, Yan Liu, and Li Hu. "Optimal frequencyhopping anti-jamming strategy based on multi-step prediction Markov decision process," Wireless Networks, vol. 27, no. 7, pp. 4581-4601, 2021.
- [65] H. Im and S.-H. Lee, "Anti-Jamming Games in Multi-Band Wireless Ad Hoc Networks," IEEE Transactions on Information Forensics and Security, vol. 18, 2023.
- [66] M.S. Abdalzaher, K. Seddik, M. Elsabrouty, O. Muta, H. Furukawa, and A. Abdel-Rahman, "Game Theory Meets Wireless Sensor Networks Security Requirements and Threats Mitigation: A Survey," Sensors, vol. 16, no. 7, pp. 1003, 2016.
- [67] M. Zhou, Y. Guan, M. Hayajneh, K. Niu, C. Abdallah, "Game Theory and Machine Learning in UAVs-Assisted Wireless Communication Networks: A Survey," Preprint submitted to Journal of LaTeX Templates, arXiv:2108.03495 [cs.MA], August 10, 2021.

- [68] J. Tsai, Y. Qian, M. Tambe, Y. Vorobeychik, and C. Kiekintveld, "Bayesian Security Games for Controlling Contagion," in Proceedings of SocialCom/ PASSAT/ BigData/ EconCom/ BioMedCom, 2013.
- [69] A. Garnaev, W. Trappe, and A. Petropulu, "Combating Jamming in Wireless Networks: A Bayesian Game with Jammer's Channel Uncertainty," in Proceedings of the IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), 2019.
- [70] F. M. Aziz, L. Li, J. S. Shamma, and G. L. Stuber, "Smart jammer and LTE network strategies in an infinite-horizon zero-sum repeated game with asymmetric and incomplete information," arXiv preprint arXiv:1904.11184, 2019.
- [71] K. Dabcevic, A. Betancourt, L. Marcenaro, and C. S. Regazzoni, "Intelligent cognitive radio jamming-a game-theoretical approach," EURASIP Journal on Advances in Signal Processing, 2014.
- [72] X. Liu, Z. Huang, Q. Wang, Y. Chen, and Y. Cao, "A Repeated Game-Based Distributed Denial of Service Attacks Mitigation Method for Mining Pools," Electronics, vol. 13, no. 2, p. 398, 2024.
- [73] J. A. Shaikh, C. Wang, M. A. Khan, S. A. H. Mohsan, S. Ullah, S. A. Chelloug,
 M. S. A. Muthanna, and A. Muthanna, "A UAV-assisted Stackelberg game model for securing LoMT healthcare networks," Drones, vol. 7, no. 7, p. 415, 2023.
- [74] Z. Wu, L. Tian, Y. Zhang, Y. Wang, and Y. Du, "Network Attack and Defense Modeling and System Security Analysis: A Novel Approach Using Stochastic Evolutionary Game Petri Net," Security and Communication Networks, vol. 2021, pp. 1-10, 2021.
- [75] D. Barik, J. Sanyal, and T. Samanta, "Denial-of-service attack mitigation in multihop 5G D2D wireless communication networks employing double auction game," Journal of Network and Systems Management, vol. 31, no. 1, pp. 1-2023, 2023.

- [76] Y. Chakhchoukh and H. Ishii, "Coordinated cyber-attacks on the measurement function in hybrid state estimation," IEEE Transactions on Power Systems, vol. 30, no. 5, pp. 2487-2497, 2014.
- [77] S. Bi and Y. J. Zhang, "Graphical methods for defense against false-data injection attacks on power system state estimation," IEEE Transactions on Smart Grid, vol. 5, no. 3, pp. 1216-1227, 2014.
- [78] N. Nikmehr and S. M. Moghadam, "Game-theoretic cybersecurity analysis for false data injection attack on networked microgrids," IET Cyber-Physical Systems: Theory & Applications, vol. 4, no. 4, pp. 365-373, 2019.
- [79] R. Zhang and P. Venkitasubramaniam, "False data injection and detection in LQG systems: A game theoretic approach," IEEE Transactions on Control of Network Systems, vol. 7, no. 1, pp. 338-348, 2019.
- [80] D. Lin, Q. Wang, and P. Yang, "The game theory: applications in the wireless networks," in Game Theory—Applications in Logistics and Economy, 2018.
- [81] Z. Han, Game Theory in Wireless and Communication Networks: Theory, Models, and Applications. Cambridge University Press, 2012.
- [82] M. Kim, E. Hwang, and J.-N. Kim, "Analysis of eavesdropping attack in mmWave-based WPANs with directional antennas," Wireless Networks, vol. 23, pp. 355-369, 2017.
- [83] M. Kim, "Game theoretic approach of eavesdropping attack in millimeter-wavebased WPANs with directional antennas," Wireless Networks, vol. 25, no. 6, pp. 3205-3222, 2019.
- [84] Y. Kwon, X. Wang, and T. Hwang, "A game with randomly distributed eavesdroppers in wireless ad hoc networks: a secrecy EE perspective," IEEE Transactions on Vehicular Technology, vol. 66, no. 11, pp. 9916-9930, 2017.
- [85] Q. Zhu and T. Başar, "Game-theoretic approach to feedback-driven multi-stage moving target defense," in International Conference on Decision and Game Theory for Security, pp. 246-263, 2013.

- [86] Y. Luo, Z. Feng, H. Jiang, Y. Yang, Y. Huang, and J. Yao, "Game-theoretic learning approaches for secure D2D communications against full-duplex active eavesdropper," IEEE Access, vol. 7, pp. 41324-41335, 2019.
- [87] Z. Han, N. Marina, M. Debbah, and A. Hjørungnes, "Physical layer security game: interaction between source, eavesdropper, and friendly jammer," EURASIP Journal on Wireless Communications and Networking, vol. 2009, pp. 1-10, 2010.
- [88] L. Jia, N. Qi, F. Chu, S. Fang, X. Wang, S. Ma, and S. Feng, "Game-theoretic learning anti-jamming approaches in wireless networks," IEEE Communications Magazine, vol. 60, no. 5, pp. 60-66, 2022.
- [89] R. K. Mallik, R. A. Scholtz, and G. P. Papavassilopoulos, "Analysis of an on-off jamming situation as a dynamic game," IEEE Transactions on Communications, vol. 48, no. 8, pp. 1360-1373, 2000.
- [90] M. Costa, Y. E. Sagduyu, T. Erpek, and M. Médard, "Robust improvement of the age of information by adaptive packet coding," in Proceedings of ICC 2021-IEEE International Conference on Communications, pp. 1-6, IEEE, 2021.
- [91] M. Costa and Y. E. Sagduyu, "Timely and covert communications under deep learning-based eavesdropping and jamming effects," Journal of Communications and Networks, vol. 25, no. 5, pp. 621-630, 2023.
- [92] T. E. Carroll and D. Grosu, "A game theoretic investigation of deception in network security," Security and Communication Networks, vol. 4, no. 10, pp. 1162-1172, 2011.
- [93] T. H. Nguyen, M. Wright, M. P. Wellman, and S. Baveja, "Multi-stage attack graph security games: Heuristic strategies, with empirical game-theoretic analysis," in Proceedings of the 2017 Workshop on Moving Target Defense, pp. 87-97, 2017.
- [94] S. Banerjee and S. Ulukus, "The freshness game: Timely communications in the presence of an adversary," arXiv preprint arXiv:2302.14024, 2023.

- [95] S. Banerjee and S. Ulukus, "Age of information in the presence of an adversary," in IEEE INFOCOM 2022-IEEE Conference on Computer Communications Workshops (INFOCOM WKSHPS), pp. 1-8, IEEE, 2022.
- [96] A. Mandal, R. Bhattacharjee, and A. Sinha, "Optimizing age-of-information in adversarial environments with channel state information," in 2022 14th International Conference on COMmunication Systems & NETworkS (COMSNETS), pp. 522-530, IEEE, 2022.
- [97] X. Sun, F. Zhao, H. H. Yang, W. Zhan, X. Wang, and T. Q. Quek, "Optimizing age of information in random-access poisson networks," IEEE Internet of Things Journal, vol. 9, no. 9, pp. 6816-6829, 2021.
- [98] A. Muhammad, M. Elhattab, M. A. Arfaoui, A. Al-Hilo, and C. Assi, "Age of information optimization in RIS-assisted wireless networks," IEEE Transactions on Network and Service Management, 2023.
- [99] R.-J. Reifert, S. Roth, and A. Sezgin, "Optimizing the Age of Information in Mixed-Critical Wireless Communication Networks," in ICC 2023-IEEE International Conference on Communications, pp. 1682-1687, IEEE, 2023.
- [100] H. Chen, Q. Wang, P. Mohapatra, and N. Pappas, "Secure status updates under eavesdropping: Age of information-based physical layer security metrics," arXiv preprint arXiv:2002.07340, 2020.
- [101] R.-J. Reifert, S. Roth, and A. Sezgin, "Optimizing the Age of Information in Mixed-Critical Wireless Communication Networks," in ICC 2023-IEEE International Conference on Communications, pp. 1682-1687, IEEE, 2023.
- [102] R. Talak, S. Karaman, and E. Modiano, "Optimizing information freshness in wireless networks under general interference constraints," in Proceedings of the Eighteenth ACM International Symposium on Mobile Ad Hoc Networking and Computing, pp. 61-70, 2018.
- [103] M. N. A. Khalid, A. A. Al-Kadhimi, and M. M. Singh, "Recent developments in game-theory approaches for the detection and defense against advanced persistent threats (APTs): a systematic review," Mathematics, vol. 11, no. 6, p. 1353, 2023.

[104] H.-Y. Shi, W.-L. Wang, N.-M. Kwok, and S.-Y. Chen, "Game theory for wireless sensor networks: a survey," Sensors, vol. 12, no. 7, pp. 9055-9097, 2012.